I-2021-002597 Book 0923 Pg 393 09/15/2021 10:00 am Pg 0393-0393 Fee \$18.00 Doc: \$0.00 Shelly Russell - Love County Clerk State of Oklahoma

RESOLUTION NO. 2021-34

A RESOLUTION FOR THE FILING AND NOTIFICATION OF THE PUBLICATION OF THE 2021 BIENNIAL SUPPLEMENT OF PENAL AND OTHER ORDINANCES.

WHEREAS, the City of Marietta has prepared the City's 2021 Biennial Supplement of Penals and Other Ordinances adopted by the City from October 8, 2019, through September 10, 2021 inclusive;

WHEREAS, the City is required every two years to publish its Biennial Supplement of Penal Ordinances and to deposit a copy of the Biennial Supplement of Penal Ordinances with the County Law Library pursuant to Sections 14-109 and 14-110 of Title 11 of the Oklahoma Statutes;

WHEREAS the City has also included in its Biennial Supplement other ordinances which have been adopted;

WHEREAS, the City adopted its current Code, as amended, on January 9, 2006; and

WHEREAS, the City is required to adopt a resolution notifying the public of the publication of its 2021 Biennial Supplement pursuant to Section 14-110 of Title 11 of the Oklahoma Statutes and to file a copy of the resolution and the Office of the County Clerk;

NOW THEREFORE be it resolved by the City Council of the City:

THAT the public is hereby notified of the publication of the 2021 Biennial Supplement of Penal and Other Ordinances, from October 8, 2019 through September 10, 2021, inclusive, and that copies of the 2021 Biennial Supplement of Penal and Other Ordinances are available for review in the Office of the City Clerk; and

THAT the City Clerk shall cause one copy of this resolution to be filed with the Office of the County Clerk of Love County; and

THAT the City Clerk shall cause one copy of this Resolution to be filed with the Law Library of Love County; and

THAT the City Clerk shall keep at least one copy of the 2021 Biennial Supplement of Penal and Other Ordinances in the Office of the City Clerk for public use, inspection and examination.

Adopted this 14th day of September, 2021 by the City Council of the City of Marietta.

BY:

Keimy 101

CITY OF MARIETTA

ACTING MAYOR/COUNCIL PRESIDENT

DOTTIE GWIN, CITY CLERK

(Seal)

Call City of Marietta 580-276-5569

CITY OF MARIETTA

BIENNIAL SUPPLEMENT

OF PENAL AND OTHER ORDINANCES

ADOPTED FROM OCTOBER 8, 2019, THROUGH SEPTEMBER 10, 2021

Table of Contents

Biennial Supplement Adopted September 14, 2021

Reference:	Page Number:
90-33 and 90-56 Ordinance 2019-09.	1
2-172 Ordinance 2019-10	3
Employee Retirement Ordinance 2019-11	5
5-5-3 (4) Appendix A Ordinance 2019-12	82
Zoning-Lot 18, Block 2 Shellenberger Addition Ordinance 2020-01	
70-400-to 70-407 And 2-174 Ordinance 2020-02	

ORDINANCE NO. 2019-09

ORDINANCE AMENDING SECTION 90-33 OF ARTICLE II AND SECTION 90-56 OF ARTICLE III, OF CHAPTER 90 OF THE MARIETTA CODE OF ORDINANCES CONTINUING THE TOTAL EXCISE TAX OF THREE PERCENT (3%); PROVIDING THAT THE INCREASE OF THREE PERCENT (3%) WILL EXPIRE IN 5 YEARS; AND ESTABLISHING REPEALER

BE IT ORDAINED BY THE CITY COUNCIL OF THE CITY OF MARIETTA, OKLAHOMA:

Section 1. Section 90-33 of Article II of Chapter 90 of the Marietta Code of Ordinances shall be amended to state as follows:

Sec. 90-33. Levy.

- (a) There is levied an excise tax of three percent (3%) upon the gross receipts derived from all sales taxable under the state sales tax code, 68 O.S. § 1350 et seq., including but not exclusive of the following:
 - (1) Tangible personal property.
 - (2) Natural or artificial gas, electricity, ice, steam or any other utility or public service except water.
 - (3) Transportation for hire of persons by common carriers, including railroads, both steam and electric, motor transportation companies, taxicab companies, pullman car companies, airlines and all other means of transportation for hire.
- (b) All other sections of this article to the contrary notwithstanding, such tax is also levied on all sales of natural or artificial gas and electricity.
- (c) The levy of excise tax of three percent (3%), will expire five (5) years from March 3, 2020, and upon such expiration, the excise tax will revert to the previous amount of two percent (2%).
- Section 2. Section 90-56 of Article III of Chapter 90 of the Marietta Code of Ordinances shall be amended to state as follows:

- Sec. 90-56. Excise tax on storage, use or other consumption of tangible, personal property levied.
- (a) There is hereby levied and there shall be paid by every person storing, using or otherwise consuming within the municipality tangible, personal property purchased or brought into this municipality, an excise tax on the storage, use or other consuming within the municipality of such property at the rate of three percent (3%) of the purchase price of such property. Such tax shall be paid by every person storing, using or otherwise consuming, within the municipality, tangible, personal property purchased or brought into the municipality. The additional tax levied hereunder shall be paid at the time of importation or storage of the property within the municipality and shall be assessed to only property purchased outside Oklahoma; provided, that the tax levied herein shall not be levied against tangible, personal property intended solely for use outside the municipality, but which is stored in the municipality pending shipment outside the municipality or which is temporarily retained in the municipality for the purpose of fabrication, repair, testing, alteration, maintenance or other service. Any person liable for payment of the tax authorized herein, may deduct from such tax any local or municipal sales tax previously paid on such goods or services; provided, that the amount deducted shall not exceed the amount that would have been due if the taxes imposed by the municipality had been levied on the sale of such goods or services.
- (b) The levy of excise tax of three percent (3%), will expire five (5) years from March 3, 2020, and upon such expiration, the excise tax will revert to the previous amount of two percent (2%).

Section 3. Any ordinances, including Ordinance 2019-08, or code provisions, or any parts thereof, in conflict with the provisions of this ordinance are repealed.

Passed by the City Council the City Marietta, Oklahoma and signed by the Mayor on this 8th day of October 2019.

CITY OF MARIETTA

ATTEST:

KERMIT MCKINNEY

ACTING MAYOR/COUNCIL PRESIDENT

Ordinance 2019-09

Y CLERK

ORDINANCE NO. 2019-10

AN ORDINANCE AMENDING SECTION 2-172 OF THE CODE OF ORDINANCES OF THE CITY OF MARIETTA ALLOW THE SALE OF SURPLUS SUPPLIES, MATERIALS OR EQUIPMENT TO ANOTHER UNIT OF GOVERNMENT WITHOUT REQUIRING AN OPPORTUNITY FOR COMPETITIVE BIDDING; REPEALING ORDINANCES OR PARTS OF ORDINANCES IN CONFLICT; PROVIDING FOR SEVERABILITY; AND DECLARING AN EMERGENCY

BE IT ORDAINED BY THE MAYOR AND THE CITY COUNCIL OF THE CITY OF MARIETTA, OKLAHOMA:

SECTION 1. Section 2-172 (b) of the Code of Ordinances of the City of Marietta which provides:

Before the sale of any surplus or obsolete supplies, materials or equipment, the value of which exceeds the sum of \$1,000.00, ample opportunity for competitive bidding shall be given. The sale shall be awarded to the highest and best responsible bidder.

is amended to read as follows:

- (b) Before the sale of any surplus or obsolete supplies, materials or equipment, the value of which exceeds the sum of \$1,000.00, ample opportunity for competitive bidding shall be given. The sale shall be awarded to the highest and best responsible bidder.
 - (1) An opportunity for competitive bidding is not required when the surplus or obsolete supplies, materials, or equipment is to be sold to another unit of government at fair market value.

SECTION 2. All ordinances or parts of ordinances in conflict herewith are, to the extent of such conflict, repealed.

SECTION 3. If any section, sub-section, sentence, clause, phrase, or portion of this Ordinance is, for any reason, held invalid or unconstitutional by any court of competent jurisdiction, such portion shall be deemed separate, distinct, and independent provision and such holding shall not affect the validity of the remaining portion of this Ordinance.

SECTION 4. It being immediately necessary for the preservation of the public peace, health, safety, and welfare of the City of Marietta and the inhabitants thereof that this ordinance be put into full effect, an emergency is hereby declared to exist by reason whereof this ordinance shall be in full force and effect from and after its passage and approval.

PASSED by the City Council and SIGNED by the Mayor of the City of Marietta, Oklahoma on this 30th day of October 2019.

ORDINANCE NO. 2019-10

CITY OF MARIETTA

KIMBERLY FRAIRE, MAYOR

ATTEST:

DOTTIE GWIN, CÎT

ORDINANCE NO. 2019-11

AN ORDINANCE AMENDING THE EMPLOYEE RETIREMENT SYSTEM, DEFINED BENEFIT PLAN FOR THE CITY OF MARIETTA, OKLAHOMA BY ADOPTING A REVISED AND RESTATED RETIREMENT PLAN; PROVIDING RETIREMENT BENEFITS FOR ELIGIBLE EMPLOYEES OF THE CITY OF MARIETTA, OKLAHOMA; PROVIDING FOR PURPOSE AND ORGANIZATION; PROVIDING FOR DEFINITIONS; PROVIDING FOR ELIGIBILITY AND PARTICIPATION; PROVIDING FOR NON-ALIENATION OF BENEFITS; PROVIDING FOR EMPLOYER AND EMPLOYEE CONTRIBUTIONS; PROVIDING FOR ACCOUNTING, ALLOCATION, AND VALUATION; PROVIDING BENEFITS; PROVIDING FOR REQUIRED NOTICE; PROVIDING FOR AMENDMENTS AND TERMINATION; PROVIDING FOR TRANSFER TO AND FROM OTHER PLANS; CREATING RETIREMENT COMMITTEE AND PROVIDING FOR POWERS, DUTIES, AND RIGHTS OF RETIREMENT COMMITTEE; PROVIDING FOR PAYMENT OF CERTAIN OBLIGATIONS; PROVIDING FOR DURATION AND PAYMENT OF EXPENSES; PROVIDING FOR EFFECTIVE DATE; PROVIDING FOR VESTING SCHEDULES; PROVIDING FOR A FUND TO FINANCE THE SYSTEM TO BE POOLED WITH OTHER INCORPORATED CITIES, TOWNS AND THEIR AGENCIES AND INSTRUMENTALITIES FOR PURPOSES OF ADMINISTRATION, MANAGEMENT, AND INVESTMENT AS PART OF THE OKLAHOMA MUNICIPAL RETIREMENT FUND: **PROVIDING** FOR **PAYMENT** OF ALL CONTRIBUTIONS UNDER THE SYSTEM TO THE OKLAHOMA MUNICIPAL) RETIREMENT FUND FOR MANAGEMENT AND INVESTMENT; PROVIDING FOR REPEALER AND SEVERABILITY; ADOPTING THOSE AMENDMENTS MANDATED BY THE INTERNAL REVENUE CODE; AND DECLARING AN EMERGENCY.

BE IT ORDAINED BY THE CITY COUNCIL OF MARIETTA, OKLAHOMA:

Section 1. That pursuant to the authority conferred by the laws of the State of Oklahoma, and for the purpose of encouraging continuity and meritorious service on the part of City employees and thereby promote public efficiency, there is hereby authorized created, established, and approved and adopted, effective as of January 1, 2020, the amended and restated Plan designated "Employee Retirement System of the City of Marietta, Oklahoma, Defined Benefit Plan," (hereinafter called System), an executed counterpart of which is marked Exhibit "A" (Joinder Agreement) and Exhibit "B" (amended and restated plan) and attached hereto as part hereof.

Section 2. FUND. A fund is hereby provided for the exclusive use and benefit of the persons, entitled to benefits under the System. All contributions to such fund shall be paid over to and received in trust for such purpose by the City. Such Fund shall be pooled for purposes of management and investment with similar funds of other incorporated cities, towns, and municipal trusts in the State of Oklahoma as a part of the Oklahoma Municipal Retirement Fund in accordance with the trust agreement of the Oklahoma Municipal Retirement Fund, a public trust. The City shall hold such contributions in the form received, and from time to time pay over and transfer the same to the Oklahoma Municipal Retirement Fund, as duly authorized and directed by the Board of Trustees. The Fund shall be nonfiscal and shall not be considered in computing any levy when the annual estimate is made to the County Excise. Board.

Section 3. APPROPRIATIONS. The City of Marietta, Oklahoma, is hereby authorized to incur, the necessary expenses for the establishment, operation, and administration of the System, and to appropriate and pay the same. In addition, the City of Marietta, Oklahoma, is hereby authorized to appropriate annually such amounts as are required in addition to employee contributions to maintain the System and the Fund in accordance with the provisions of the Defined Benefit Plan. Any appropriation so made to maintain the System and Fund shall be for deferred wages or salaries, and for the payment of necessary expenses of operation and administration to be transferred to the trustees of the Oklahoma Municipal Retirement Fund for such purposes and shall be paid into the Fund when available, to be duly transferred to the Oklahoma Municipal Retirement Fund.

Section 4. EXECUTION. The Mayor and City Clerk be and they are each hereby authorized and directed to execute (in counterparts, each of which shall constitute an original) the System instrument, and to do all other acts and things necessary, advisable, and proper to put said System and related trust into full force and effect, and to make such changes therein as may be necessary to qualify the same under Sections 401(a) and 501(a) of the Internal Revenue Code of the United States. The counterpart attached hereto as Exhibit "A" and Exhibit "B", which has been duly executed as aforesaid simultaneously with the passage of this Ordinance and made a part hereof, is hereby ratified and confirmed in all respects.

This Committee is hereby authorized and directed to proceed immediately on behalf of the City of Marietta, Oklahoma, to pool and combine the Fund into the Oklahoma Municipal Retirement Fund as a part thereof, with similar funds of such other cities and towns, for purposes of pooled management and investment.

Section 5. REPEALER. Any Ordinance inconsistent with the terms and provisions of this Ordinance is hereby repealed, provided, however, that such repeal shall be only to the extent of such inconsistency and in all other respects this Ordinance shall be cumulative of other ordinances regulating and governing the subject matter covered by this Ordinance.

Section 6. SEVERABILITY. If, regardless of cause, any section, subsection, paragraph, sentence or clause of this Ordinance, including the System as set forth in Exhibit "A" and Exhibit "B", is held invalid or to be unconstitutional, the remaining sections, subsections, paragraphs, sentences, or clauses shall continue in full force and effect and shall be construed thereafter as being the entirel provisions of this Ordinance.

Section 7. EMERGENCY. Whereas in the judgment of the City Council of the City of Marietta, Oklahoma, the public peace, health, safety, and verifare of the City of Marietta, Oklahoma, and the inhabitants thereof dentand the immediate passage of this Ordinance, an emergency is hereby declared, the rules are suspended, and this Ordinance shall be in full force and effective on its passage, approvals and publication.

END

The undersigned hereby certifies that the foregoing Ordinance was introduced before the City Council of the City of Marietta on the 12 day of November , 20 19 , and was duly adopted and approved by the Mayor and City Council, on the 12 day of November , 20 19 , after compliance with notice requirements of the Open Meeting Law (25 OSA, Sections 301, et. seq.).

City of Marietta

11120

Clerk

, 2019.

OKLAHOMA MUNICIPAL RETIREMENT FUND MASTER DEFINED BENEFIT PLAN JOINDER AGREEMENT

City of Marietta [a municipality chartered, incorporated or formed under the laws of Oklahoma], a city, town, agency, instrumentality, or public trust located in the State of Oklahoma, with its principal office at Marietta, Oklahoma, hereby establishes a Defined Benefit Plan to be known as City of Marietta Plan (the "Plan") in the form of The Oklahoma Municipal Retirement Fund Master Defined Benefit Plan.

Except as otherwise provided herein, the definitions in Article II of the Plan apply. 1. Dates. This instrument is a new Plan effective ___ [such date may not be earlier than the first day of the []Plan Year in which it is executed]. [X] This instrument is an amendment, restatement and continuation of the Previous Plan, which was originally July 1, 1981. The effective date of this Joinder Agreement is January 1, 2020 [such date may not be earlier than the first day of the Plan Year beginning in 2008, or the first day of the initial Plan Year, if later], except as otherwise stated in the Plan and the Joinder Agreement. 2. Employee. The word "Employee" shall mean: [X] Any person other than a Leased Employee who, on or after the Effective Date, is considered to be a regular full-time employee in accordance with the Employer's standard personnel policies and practices, and is receiving remuneration for such services rendered to the Employer (including any elected official and any appointed officer or employee of any department of the Employer, whether governmental or proprietary in nature), including persons on Authorized Leave of Absence. Employees shall not include independent contractors. Elected members of the City Council shall not be considered to be Employees solely by reason of their holding such office. Any person other than a Leased Employee who, on or after the Effective Date, is considered to be [] a regular employee in accordance with the Employer's standard personnel policies and practices (including part-time, seasonal and temporary employees), and is receiving remuneration for such services rendered to the Employer (including any elected official and any appointed officer or employee of any department of the Employer, whether governmental or proprietary in nature), including persons on Authorized Leave of Absence. Employees shall not include independent contractors. Elected members of the City Council shall not be considered to be Employees solely by reason of their holding such office. Any person who, on or after the Effective Date, is an employee of the Employer and holds the position of: City Manager, City or Town Administrator, President, Chief Executive Officer, General [] Manager, or District Manager, as applicable. Assistant City Manager [] Chief of Police Fire Chief [] Department Head or Department Manager Finance Director or Chief Financial Officer General Counsel or Municipal Attorney

(specify position) [do not specify the name of the individual or a finite group unless the

Plan otherwise provides continuing eligibility to a specified position or group]

Municipal Judge

[]

system approved by the City: [] City Manager, City or Town Administrator, President, Chief Executive Officer, General Manager, or District Manager, as applicable. [] Assistant City Manager [] Chief of Police [] Fire Chief [] Department Head or Department Manager [] Finance Director or Chief Financial Officer [] General Counsel or Municipal Attorney [] Municipal Judge [] (specify position) [do not specify the name of the individual or a finite group unless the Plan otherwise provides continuing eligibility to a specified position or group] [] Any person who is 3. Eligibility. Eligible Employees shall commence participation in the Plan: (Select only one) [] month(s) (any number of months up to twelve consecutive) after the Employee's Employment Commencement Date or the date the individual meets the definition of Employee in Section 2			word "Employee" shall not include: Any person who is currently accruing benefits under any other state or local retirement system.
 3. Eligibility.			 [] City Manager, City or Town Administrator, President, Chief Executive Officer, General Manager, or District Manager, as applicable. [] Assistant City Manager [] Chief of Police [] Fire Chief [] Department Head or Department Manager [] Finance Director or Chief Financial Officer [] General Counsel or Municipal Attorney [] Municipal Judge [] (specify position) [do not specify the name of the individual or a finite group unless the Plan otherwise provides continuing eligibility to a specified position or group]
Eligible Employees shall commence participation in the Plan: (Select only one) []		[]	Any person who is
Compensation shall exclude the item(s) listed below: [] No exclusions. [] Overtime pay. [] Bonuses. [] Commissions. [] Longevity pay. [X] Severance pay. [X] Accrued vacation or sick leave paid upon termination of employment and moving expenses. [] Fringe benefits, expense reimbursements, deferred compensation and welfare benefits. [] Other: [must be definitely determinable] 5. Average Monthly Compensation. The considered period for purposes of the definition of "Average Monthly Compensation" in Section 2. of the Plan is: [X] sixty (60) consecutive months.	3.	Eligil	ble Employees shall commence participation in the Plan: (Select only one) month(s) (any number of months up to twelve consecutive) after the Employee's Employmen Commencement Date or the date the individual meets the definition of Employee in Section 2 hereof, provided that the individual has met the definition of Employee in Section 2 hereo throughout such period.
The considered period for purposes of the definition of "Average Monthly Compensation" in Section 2. of the Plan is: [X] sixty (60) consecutive months.	4.	Comp [] [] [] [] [] [X] [X]	Pensation shall exclude the item(s) listed below: No exclusions. Overtime pay. Bonuses. Commissions. Longevity pay. Severance pay. Accrued vacation or sick leave paid upon termination of employment and moving expenses. Fringe benefits, expense reimbursements, deferred compensation and welfare benefits.
	5.	The of the [X]	considered period for purposes of the definition of "Average Monthly Compensation" in Section 2. e Plan is: sixty (60) consecutive months.

- 6. The Employer hereby elects the following Plan design:
 - [X] Mandatory Contribution Option. A Participant shall be required to contribute to the Plan for each Plan Year the percentage of his Compensation ("Mandatory Contributions") required by the Plan in Section 8 of this Joinder Agreement. Mandatory Contributions shall be made by payroll deductions. A Participant shall authorize such deductions in writing on forms approved by, and filed with, the Committee.

If the Participant's Mandatory Contributions pursuant to the preceding paragraph are to be taxed deferred:

- [X] <u>Pick Up Option</u>. The Employer hereby elects to have the provisions of Section 3.4 of the Plan apply. The Employer shall pick up and pay the percentage of each Participant's, Compensation required to be contributed as of <u>July 1, 1998</u> [insert date] in lieu of contributions by the Participant. No Participant shall have the option of receiving the contributed amounts directly as Compensation.
- Non-Contributory Option. Participants shall not be required nor permitted to contribute to the Plan.
- 7. A. Payment Options. The Employer hereby elects the following minimum number of payments for employees eligible to receive benefits under Article IV of the Plan:

[X] Sixty (60) monthly payments.

[] One hundred and twenty (120) monthly payments.

B. Plan Options. The Employer hereby elects the following plan designation and percentage used in calculating benefits under Section 5.1 of the Plan.

[] Plan AAA – 3.00% with no maximum Years of Service

Plan AAA – 3.00% recognizing a maximum of 22 Years of Service

[X] Plan AA 2.625%

- [] Plan BB 2,25%
- [] Plan CC 1.875%
- [] Plan A 1.50%
- [] Plan B 1.125%
- [] Plan C .75%
- C. Normal Retirement Age. Normal retirement age shall be:
 - [X] Age 65
 - [] The earlier of (i) and (ii) as follows:
 - (i) age 65
 - (ii) the later of age 62 and the age at which the Participant has completed 30 Years of Service.

Examples: An employee hired at age 20 who worked for 30 years and terminated at age 50 would be entitled to unreduced payments at age 62.

An employee hired at age 30 who worked for 25 years and terminated at age 55 would be entitled to unreduced payments at age 65.

[] Modified Rule of 80:

The earlier of (i) and (ii) as follows:

- (i) age 65
- (ii) the later of age 55 and the age at which the sum of the Participant's age in completed years and the participant's number of completed Years of Service total 80 or greater. To be eligible, the Participant's age plus Years of Service must be at least 80 prior to termination of employment (or, after termination of employment in the case of a Participant who transfers to another Municipality in accordance with Section 8.1(b) of the Plan)

Examples: 1. An employee hired at age 30 who worked for 25 years and terminated at age 55 would be entitled to unreduced payments immediately. Age 55 plus 25 years equals 80.

- 2. An employee hired at age 20 who worked for 30 years and terminated at age 50 would be entitled to unreduced payments at age 55. The employee has age plus Years of Service points at age 50 but the minimum age for payment is 55.
- 3. An employee hired at age 25 who worked for 25 years and terminated at age 50 would be entitled to unreduced payments at age 65. Age 50 plus 25 years is less than 80, so the Normal Retirement Age is 65.
- D. Vesting Options. The Employer hereby elects the following vesting option to determine an Employee's eligibility to receive retirement benefits. Ten Year Cliff Vesting Schedule [X] Seven Year Cliff Vesting Schedule Five Year Cliff Vesting Schedule E. Service Credit Prior to Effective Date. The Employer hereby elects to include the following limitation of service prior to the effective date. [X] No limitation [X] For all purposes under the Plan With respect to Service for purposes of vesting and attainment of Normal Retirement [] Service credit prior to the effective date shall not exceed years []For all purposes under the Plan [] With respect to Service for purposes of benefit accruals. F. Service Buyback. The Employer hereby elects No service buyback pursuant to Section 10.11 of the Plan [X] The service buyback provisions of Section 10.11 of the Plan. G. Service for Worker's Compensation Period. If a Participant is on an Authorized Leave of Absence and is receiving worker's compensation during such Authorized Leave of Absence, such **Participant** [X] shall be credited with Service for such period for purposes of vesting only and not for purposes of benefits, but no Employee contributions shall be made with respect to the Participant for such period. shall not be credited with Service for such period. []8. Contributions by Participants. If Participants are required to contribute to the cost of providing benefits under this Plan, such contributions shall be based on the plan designation selected in Section 7B above and shall apply to pay periods commencing on and after July 1, 2006. The Participant contribution formula in Section 3.3 of the Plan shall use the following percentage for the Plan Option selected in Section 7B of this Joinder Agreement: Pian AAA – 6.00% Plan AA - 5.25% Plan BB - 4.50% Plan CC - 3.75% Plan A - 3.00% Plan B - 2.25%

(Participant plus Employer percentages must total 100%.)

Employer as follows: Employee portion <u>%</u> Employer portion <u>%</u>

Plan C - 1.50%

b. [] c. []

The contribution will be actuarially determined based on Plan assets and liabilities as of January 1 of each year as a percent of payroll, which will then be shared between the Employer and Participant as noted above. These contribution rates will be in effect from July 1 of that year until June 30 of the subsequent year.

The contribution formula shall be $\frac{9}{2}$ [insert number between 0 and twelve] of compensation.

The contribution as annually determined each year shall be shared by the Participant and

9. Cost-of-Living Option.

For purposes of adjusting retiree and beneficiary pensions, the Employer hereby elects the following: [X] No Cost-of-Living Option.

[] Cost-of-Living Option. This election applies to Sections 5.1 (Normal Pension), 5.2 (Early Pension), 5.3 (Disability Pension), 5.4 (Deferred Vested Pension), 6.2 (Death Prior to Commencement of Pension), 6.3(a) and 6.3(b) (Death After Commencement of Pension), and 6.4 (Spouse's Pension) and provides annual benefit increases of the smaller of three percent (3%) or the percentage change in the Consumer Price Index.

The effective date of the Cost-Of-Living Option shall be, the original date that the Employer elected the Cost-Of-Living Option.

10. Retiree Plan Improvement Option.

Benefits payable to or on behalf of a former Employee under Article V, Article VI, or Article VII of the Plan, which are due or in the course of payment on or after the Effective Date of this Joinder Agreement, shall

- [] be increased according to the Plan Option elected herein. Such increased benefits shall be reflected in any periodic payments due or paid on or after the Effective Date of the Joinder Agreement. It is not intended for this change to be retroactive and any periodic payments due prior to such date shall not be affected.
- be increased by <u>%</u> effective. Such increased benefits shall be reflected in any periodic payments due or paid after such date. It is not intended for this change to be retroactive and any periodic payments due prior to such date shall not be affected.
- [X] not be increased unless such former Employee is subject to Section 10.9 or 10.10 of the Plan, but shall continue to be paid under the terms of the Previous Plan.

11. Limitations on Optional Benefit Forms.

Section 7.2 of the Plan provides for a lump sum payment form, an installment payment form that would be payable over a fixed number of years (at which time all payments would cease), or the purchase of an insured annuity. The Employer hereby elects the following:

- [X] Optional benefit forms under Section 7.2 of the Plan will not be permitted.
- Optional benefit forms under Section 7.2 of the Plan will be permitted, subject to Retirement Committee approval for any such elections by a Participant, subject to the following limitation(s): (The above election has no effect on the joint and survivor optional benefit forms under Section 7.1).

12. Defined Contribution Option.

- [X] Not applicable.
- Participant shall be entitled to the benefit under this option, in addition to the benefit determined according to Section 7B.

An account shall be created for each active Participant as of the effective date of the option. The beginning balance of the account shall be the Participant's Contribution Accumulation. The account shall be credited with:

- (I) Mandatory Contributions made by the Participant after the effective date of the option; and
- (2) Investment earnings at same rate as earned by the Oklahoma Municipal Retirement Fund (OMRF) Defined Benefit Fund.

As soon as administratively possible after termination of employment or death, the administrator shall pay the Participant or Beneficiary if applicable, the account balance as requested. The Participant may elect to receive the benefit in any of the Benefit options permitted under the plan. The benefit shall be the Actuarial Equivalent of the account balance at the time the benefit commences.

This option shall be effective [include the earlier of the date this Option was originally adopted in a Joinder Agreement or the date of adoption in the current Plan Year].

IN WITNESS WHEREOF the City of Marietta has caused its corporate seal to be affixed hereto and this instrument to be duly executed in its name and behalf by its duly authorized officers this City of Marietta Title Title: (SEAL) The foregoing Joinder Agreement is hereby approved by the Oklahoma Municipal Retirement Fund this 2 day of november, 2019. OKLAHOMA MUNICIPAL RETIREMENT FUND Attest: Required Disclosures. This Joinder Agreement is to be used only with the Oklahoma Municipal Retirement Fund Master Defined Benefit Plan. Failure to properly complete this Joinder Agreement may result in failure of the Plan to qualify under Code Section 401(a). In accordance with IRS Rev. Proc. 2011-49, the Volume Submitter Practitioner who has obtained Internal Revenue Service approval of the Oklahoma Municipal Retirement Fund Master Defined Benefit Plan has authority under the Plan document to amend the Plan on behalf of adopting employers for certain changes in the Code, regulations, revenue rulings, other statements published by the Internal Revenue Service, including model, sample or other required good faith amendments. The Volume Submitter Practitioner will inform adopting employers of any such amendments or of the discontinuance or abandonment of the volume submitter plan document. The name, address and telephone number of the Volume Submitter Practitioner are: McAfee & Taft A Professional Corporation, 10th Floor, Two Leadership Square, 211 N. Robinson, Oklahoma City, OK 73102, telephone (405) 552-2231. Any inquiries by the adopting employer regarding the adoption of the Plan, the meaning of Plan provisions, or the effect of the Internal Revenue Service advisory letter on

13. The Employer has consulted with and been advised by its attorney concerning the meaning of the

provisions of the Plan and the effect of entry into the Plan.

the volume submitter plan may be directed to the Volume Submitter Practitioner.

OKLAHOMA MUNICIPAL RETIREMENT FUND MASTER DEFINED BENEFIT PLAN

OKLAHOMA MUNICIPAL RETIREMENT FUND MASTER DEFINED BENEFIT PLAN

TABLE OF CONTENTS

		<u>Page</u>
ART	ICLE I. Pui	rpose and OrganizationI-1
1.1		
1.2		I-1
ART.	ICLE II. De	efinitions and ConstructionII-1
2.1	Definitio	nsII-1
	(a)	Accrued Pension II-1
	(b)	Actuarial Equivalent II-1
	(c)	Adjustment Factor II-1
	(d)	Authorized AgentII-1
	(e)	Authorized Leave of Absence II-1
	(f)	Average Monthly Compensation II-1
	(g)	BeneficiaryII-2
	(h)	Break in ServiceII-2
	(i)	City CouncilII-2
	(j)	Code II-2
	(k)	Committee II-2
	(1)	CompensationII-2
	(m)	Contribution Accumulation
	(n)	Death BenefitII-3
	(o)	Deferred Vested PensionII-3
	(p)	Disability II-3
	(q)	Disability Pension II-3
	(r)	Early PensionII-3
	(s)	Effective Date II-3
	(t)	Employer II-4
	(u)	Employment Commencement Date II-4
	(v)	Fund II-4
	(w)	Joinder AgreementII-4
	(x)	Leased EmployeeII-4
	(y)	Limitation Compensation
	(z)	Municipality II-4
	(aa)	Normal Pension II-4
	(bb)	Normal Retirement Date II-4
	(cc)	Oklahoma Municipal Retirement FundII-4
	(dd)	ParticipantII-5
	(ee)	Pension II-5
	(ff)	Plan II-5
	(gg)	Plan AdministratorII-5
	(nh)	Plan Year

	(ii) Previous Plan	II-5
	(jj) Retirement	II-5
	(kk) Service	II-5
	(II) Severance Benefit	II-6
	(mm) Spouse	II-6
	(nn) Trust Service Provider	
	(00) Trustee	
	(pp) U. S. Consumer Price Index	
	(qq) Year of Service	
2.2	Construction	
ART:	ICLE III. Contributions	III-1
3.1	Eligibility	
3.2	Contributions by Employer	
3.3	Contributions by Participants	
3.4	Pick-up Contributions	
3.5	Transfer of Contributions.	
-		
	ICLE IV. Requirements for Retirement Benefits	
4.1	Normal Pension	
4.2	Early Pension	
4.3	Disability Pension	IV-1
4.4	Deferred Vested Pension	IV-2
4.5	Pensions for Former Employees	IV-2
4.6	Termination of Employment - Vesting of Benefits	IV-2
ART	ICLE V. Amount of Retirement Benefits	V-1
5.1	Normal Pension	V-1
5.2	Early Pension	
5.3	Disability Pension	
5.4	Deferred Vested Pension	
5.5	Accrued Credits and Vested Benefits Preserved	
ART	ICLE VI. Severance and Death Benefits	
6.1	Severance Benefit	VI-1
6.2	Death Prior to Commencement of Pension	
6.3	Death After Commencement of Pension	
6.4	Spouse's Pension	
6.5	Designation of Beneficiary	
6.6	Severance of Death Benefits for Former Employees	
6.7	HEART Act Provision	
ART	ICLE VII. Optional Retirement Benefits	
7.1	Joint and Survivor Options	VII-1
7.2	Other Forms of Payment	
7.3	Restrictions on Optional Forms	
7.4	Other Benefits Canceled by Ontion	

7.5	Options by Former Employee	VII-3
7.6	Rollover to Another Plan or IRA	
7.7	Minimum Distribution Requirements	VII-4
ARTI	CLE VIII. Employment Transfers	VIII-1
8.1	Transfers From This Plan	VIII-1
8.2	Transfers to This Plan	VIII-1
8.3	Notice of Transfers	
ARTI	CLE IX. Administration	IX-1
9.1	Administration	IX-1
9.2	Bonds	IX-4
9.3	Benefit Payments	IX-4
9.4	Abandonment of Benefits	IX-4
9.5	Benefits Payable to Incompetents	IX-4
9.6	No Participant Loans Permitted	
ARTIC	CLE X. Limitations	X-1
10.1	Limitations on Benefits Relating to Section 415 of Internal Revenue Code of 19	86 X-1
10.2	Definitions	X-1
10.3	Predecessor Employer	X-10
10.4	Severance from Employment	X-10
10.5	Year of Participation	
10.6	Year of Service	
10.7	Other Rules:	
10.8	Participant Limitation Applicable to Deferred Contribution Option	
10.9	Re-employment of Former Employees	
10.10	Re-employment of Retired Participants	
10.11	Buyback of Service	
10.12	Loss of Benefits	
10.13	Loss of Benefits for Cause	
ARTI	CLE XI. Guarantees and Liabilities	XI-1
11.1	Non-Guarantee of Employment	XI-1
11.2	Rights to Fund Assets	
11.3	Non-Alienation of Benefits	
11.4	Disclaimer of Liability	
11.5	Indemnification of Trustees	XI-1
11.6	Payments Under a Qualified Domestic Relations Order:	
ARTI	CLE XII. Amendments	XII-1
12.1	Right to Amend	XII-1
12.2	Amendments	
12.3	Authority of Volume Submitter Practitioner to Amend for Adopting Employers	XII-1
ARTI	CLE XIII. Termination	
13.1	Right to Terminate	XIII-1

13.2	Liquidation of Fund	XIII-1
13.3	Manner of Distribution	
13.4	Consolidation or Merger	XIII-2
13.5	Limitations	
ARTIC	CLE XIV. General	XIV-1
14.1	USERRA	XIV-1
14.2	Not Contract Between Employer and Participant	
14.3	Payment of Fees	XIV-1
14.4	Governing Law	
14.5	Counterpart Execution	XIV-1
14.6	Severability	XIV-1
14.7	Number and Gender	
14.8	Compensation and Expenses of Administration	XIV-1
14.9	Incorporation of Trust Agreement	
14.10		

ARTICLE I.

Purpose and Organization

1.1 Purpose: The purpose of this Plan is to encourage the loyalty and continuity of service of the Participants, to provide retirement benefits for all eligible Employees of the Employer, as hereinafter defined, who complete a period of faithful service and become eligible hereunder, and to qualify the Plan under Sections 401(a) and 501(a) of the Code by meeting the requirements of Code Section 414(d). The benefits provided by this Plan will be paid from a Fund established by the Employer and will be in addition to the benefits Employees are entitled to receive under any other programs of the Employer and from the Federal Social Security Act.

This Plan and the separate related Fund forming a part hereof are established and shall be maintained for the exclusive benefit of the eligible Employees of the Employer and their Beneficiaries. To the extent this Plan is a governmental retiree benefit plan under Section 401(a)(24) of the Code, and prior to the termination of the Plan and satisfaction of all liabilities of the Plan, no part of the corpus or income of the Fund shall be used for, or diverted to, purposes other than for the exclusive benefit of the Plan participants and their beneficiaries.

1.2 Parties: The Oklahoma Municipal Retirement Fund hereby adopts and establishes this Plan for the benefit of Employees of those Employers, as defined herein, formed, chartered or incorporated under the laws of the State of Oklahoma, who wish to adopt it by executing a Joinder Agreement which incorporates this Plan by reference.

ARTICLE II.

Definitions and Construction

- 2.1 **Definitions:** Where the following words and phrases appear in this Plan, they shall have the respective meanings set forth below, unless their context clearly indicates to the contrary:
- (a) Accrued Pension: The Pension (other than a Disability Pension) determined under the Plan expressed in the form of a monthly benefit commencing at Normal Retirement Date (or date of determination in the case of a Late Pension), which an Employee has accrued at any time under the provisions of the Plan, regardless of his vested status, determined as if he had then terminated employment.
- (b) Actuarial Equivalent: Equality in value of the aggregate amounts expected to be received under different forms of payment. Except as otherwise specifically noted, the determination of such equality will be based on the use of the 1983 Group Annuity Mortality (GAM-83) Table and 7½% interest. For purposes of determining the benefit limitations under §415(b)(2)(B), (C), or (D) of the Internal Revenue Code as set forth in Section 10.1 of the Plan the applicable mortality table for annuity starting dates prior to December 31, 2002 is set forth in Rev. Rul. 95-6. 1995-1 C.B. 80, and for annuity starting dates on or after December 31, 2002 is set forth in Rev. Rul. 2001-62, 2001-53, I.R.B. 632.
- (c) Adjustment Factor: The words "Adjustment Factor" shall mean the cost of living adjustment factor prescribed by the Secretary of the Treasury under Section 415(d) of the Code for years beginning after December 31, 1987, as applied to such items and in such manner as the Secretary shall provide.
- (d) Authorized Agent: The City Clerk of the Employer or such other person designated by the Employer to carry out the efficient operation of the Plan at the local level.
- (e) Authorized Leave of Absence: Any absence authorized by the Employer under the Employer's standard personnel practices applied to all persons under similar circumstances in a uniform manner, including any required military service during which a Participant's reemployment rights are protected by law; provided that he resumes employment with the Employer within the applicable time period established by the Employer or by law.
- Compensation paid to an Employee during a considered period by the number of months, including fractional months, for which such Compensation was received. The considered period shall be the number of consecutive months of reported pay selected in Joinder Agreement within the last one hundred twenty (120) months of service which yield the highest average Compensation. For purposes of determining consecutive months, periods of credited service shall be bridged, if interrupted with non-credited periods under an Authorized Leave of Absence. If an Employee has less than the number of months of consecutive employment service selected in the Joinder Agreement, the Employee's actual consecutive months shall be the basis for calculating the Employee's Average Monthly Compensation hereunder.

- ' (g) Beneficiary: Any person or entity designated or deemed designated by a Participant as provided in Section 6.5 hereof.
- (h) Break in Service: The expiration of ninety (90) days from the date the Participant last performed Service for the Employer for which such Participant was entitled to wages as defined in Section 3121(a) of the Code, unless the Participant is on Authorized Leave of Absence. If an Employee does not resume employment with the Employer upon the expiration of an Authorized Leave of Absence, the Participant will be deemed to be absent from work on the first day of his Authorized Leave of Absence for purposes of determining if the Participant has a Break in Service.
- (i) City Council: The City Council (or Board of Trustees) of the Employer or other duly qualified and acting governing authority of the Employer.
 - (j) Code: The Internal Revenue Code of 1986, as amended from time to time.
- (k) Committee: The City Council of the Municipality, which shall act as the Plan Administrator of the Plan as provided for under Article IX hereof.
- (I) Compensation: Compensation means wages for federal income tax withholding purposes, as defined under Code §3401(a), plus all other payments to an Employee in the course of the Employer's trade or business, for which the Employer must furnish the Employee a written statement under Code §\$6041, 6051 and 6052, but determined without regard to any rules that limit the remuneration included in wages based on the nature or location of the employment or services performed (such as the exception for agricultural labor in Code §3401(a)(2)). The Employer in Section 4 of its Joinder Agreement may specify modifications to the definition of Compensation, for purposes of benefit accruals, the calculation of benefits, or contribution allocations under the Plan. For purposes of determining an Employee's compensation, any election by such Employee to reduce his regular cash remuneration under Code Sections 125, 402(e)(3), 402(h), 403(b) or 132(f) shall be disregarded.

For Plan Years beginning after December 31, 2008, (i) an individual receiving a differential wage payment, as defined by Code Section 3401(h)(2), shall be treated as an Employee of the Employer making the payment, (ii) the differential wage payment shall be treated as Compensation, and (iii) the Plan shall not be treated as failing to meet the requirements of any provision described in Code Section 414(u)(1)(C) by reason of any contribution or benefit which is based on the differential wage payment.

Limitations. Notwithstanding anything herein to the contrary, for Plan Years commencing after December 31, 1988 and before January 1, 1994, the annual Compensation of each Participant taken into account under the Plan for any Plan Year shall not exceed \$200,000, as adjusted by the Secretary at the same time and in the same manner as under Section 415(d) of the Code except that the dollar increase in effect on January 1 of any calendar year is effective for Plan Years beginning in such calendar year and the first adjustment to the \$200,000 limitation is effective on January 1, 1990. For years beginning on or after January 1, 1994, the annual compensation limit of each Participant taken into account for determining all benefits provided under the Plan for any determination period shall not exceed \$150,000, as adjusted for the cost-of-living in accordance

with section 401(a)(17)(B) of the Internal Revenue Code. The cost-of-living adjustment in effect for a calendar year applies to any determination period beginning in such calendar year. If a determination period consists of fewer than 12 months, the OBRA '93 annual compensation limit will be multiplied by a fraction, the numerator of which is the number of months in the determination period, and the denominator of which is 12.

The annual Compensation of each Participant taken into account in determining benefit accruals in any Plan Year beginning after December 31, 2001, shall not exceed \$200,000. Annual Compensation means Compensation during the Plan Year or such other consecutive 12-month period over which Compensation is otherwise determined under the Plan (the determination period). For purposes of determining benefit accruals in a Plan Year beginning after December 31, 2001, Compensation for any prior determination period shall be \$200,000. The \$200,000 limit on annual Compensation shall be adjusted for cost-of-living increases in accordance with Section 401(a)(17)(B) of the Code. The cost-of-living adjustment in effect for a calendar year applies to annual Compensation for the determination period that begins with or within such calendar year.

If Compensation for any prior determination period is taken into account in determining an employee's benefits accruing in the current Plan Year, the Compensation for that prior determination period is subject to the applicable annual compensation limit in effect for that prior determination period.

- (m) Contribution Accumulation: The Employee's aggregate contributions, plus interest thereon accrued at the rate determined by the Trustee, compounded according to uniform rules adopted by the Trustees. Prior to January 1, 1983 the interest rate for crediting interest was three and one-half percent (3½%) per annum. In the event that the Employer has elected the Defined Contribution Option in Section 12 of the Joinder Agreement, then, as of the effective date of such election, the interest rate for determining the investment earnings on such amounts contributed under such option shall be equal to the rate earned by the Fund.
 - (n) Death Benefit: The pension benefit described in Article VI herein.
- (o) Deferred Vested Pension: The pension benefit described in Sections 4.4 and 5.4 herein.
- (p) Disability: A physical or mental condition which, in the judgment of the Committee, totally and presumably permanently prevents an Employee from engaging in any substantial gainful employment with the Employer. A determination of such disability shall be based upon competent medical evidence.
 - (q) Disability Pension: The pension benefit described in Sections 4.3 and 5.3 herein.
 - (r) Early Pension: The pension benefit described in Sections 4.2 and 5.2 herein.
- (s) Effective Date: The later of (i) the date specified in the Joinder Agreement, or (ii) the first day on which the Plan has a Participant.

- (t) Employer: A Municipality located in the State of Oklahoma which executes the Joinder Agreement.
- (u) Employment Commencement Date: The date on which the Employee's most recent employment with the Employer began.
- (v) Fund: The fund established to provide the benefits under the Plan for the exclusive benefit of the employees included in the Plan, and which will be pooled with similar funds of other incorporated cities and towns of Oklahoma as a part of Oklahoma Municipal Retirement Fund, for purposes of pooled management and investment.
- (w) Joinder Agreement: The agreement by which the Employer adopts this Plan and Fund as its Plan and Fund.
- (x) Leased Employee: Any person (other than an employee of the recipient) who pursuant to an agreement between the recipient and any other person ("leasing organization") has performed services for the recipient (or for the recipient and related persons determined in accordance with Section 414(n)(6) of the Code) on a substantially full time basis for a period of at least one year, and such services are performed under primary direction or control by the recipient. Contributions or benefits provided a leased employee by the leasing organization which are attributable to services performed for the recipient employer shall be treated as provided by the recipient employer.

A leased employee shall not be considered an employee of the recipient if: (I) such employee is covered by a money purchase pension plan providing: (1) a nonintegrated employer contribution rate of at least 10% of compensation, as defined in Section 415(c)(3) of the Code, but including amounts contributed pursuant to a salary reduction agreement which are excludable from the employee's gross income under Section 125, Section 402(e)(3), Section 402(h)(1)(B) or Section 403(b) of the Code, (2) immediate participation, and (3) full and immediate vesting; and (ii) leased employees do not constitute more than 20% of the recipient's nonhighly compensated work force.

- (y) Limitation Compensation: Compensation as defined in Section 10.2(b) hereof.
- (z) Municipality: (1) each and every municipality located in the State of Oklahoma; (2) public trusts having municipalities as Beneficiaries; (3) interlocal cooperatives between municipalities and/or their public trust, and; (4) any other legal entity comprising a municipal authority which has adopted the Plan and/or which has become a Participant in the related trust according to the terms hereof.
 - (aa) Normal Pension: The pension benefit described in Sections 4.1 and 5.1 herein.
- (bb) Normal Retirement Date: The later of (i) the Effective Date, or (ii) the first day of the month coincident with or next following the later of the Normal Retirement Age as designated in the Joinder Agreement, Section 7C, and (iii) the date he has satisfied the vesting requirements specified in the Joinder Agreement to become 100% vested.
- (cc) Oklahoma Municipal Retirement Fund: The entity known as the Oklahoma Municipal Retirement Fund, which was created to combine pension and retirement funds of

Oklahoma cities and towns for purposes of management and investment, represented by and acting through its Board of Trustees.

- (dd) Participant: Any Employee or former Employee who meets the eligibility requirements and is covered under the Plan.
- (ee) Pension: A series of monthly amounts which are payable to a person who is entitled to receive benefits under the Plan.
- (ff) Plan: The Oklahoma Municipal Retirement Fund Master Defined Benefit Plan set forth herein, and all subsequent amendments.
- (gg) Plan Administrator: The persons who administer the Plan pursuant to the provisions of Article IX hereof.
- (hh) Plan Year: The twelve (12) consecutive month period ending June 30th of each year. The initial or final Plan Year may be less than a twelve (12) consecutive month period.
- (ii) Previous Plan: The terms and provisions in the prior instruments governing the Employer's qualified defined benefit retirement plan and related trust, and applying before the Effective Date hereof, or any other date expressly specified herein if different from the Effective Date, which prior instruments are amended, restated and superseded by this instrument.
- (ij) Retirement: Termination of employment after a Participant has fulfilled all requirements for a Pension and has attained age 55 or older. Retirement shall be considered as commencing on the day immediately following an Employee's last day of employment.

(kk) Service:

- (1) A Participant's last continuous period during which the Participant was an Employee of the Employer and/or any other Municipality prior to the earlier of his retirement or Break in Service.
 - (i) Service includes employment with a Municipality other than the Employer prior to the time that the other Municipality adopted the Plan if the other Municipality credits a Participant's past service under its retirement plan; and
 - (ii) Service for the Employer does not include employment with any Municipality if that service would not be included under the Municipality's retirement plan.
- (2) Concurrent employment with more than one Municipality shall be credited as only one period of Service.
- (3) Any Authorized Leave of Absence shall not be considered as interrupting continuity of employment, provided the Employee returns within the period of authorized absence. Until such time as the City Council shall adopt rules to the contrary, credit for Service with the Employer shall be granted for any period of Authorized Leave of Absence during which the Employee's full Compensation is continued and contributions to the Fund are continued at the same

rate and made by or for him, but credit for Service with the Employer shall not be granted for any period of authorized, nonpaid absence due to illness, union leave, military service, or any other reason, unless arrangements are made with the City Council for the Employee's continued participation and for contributions to be continued at the same rate and made by him or on his behalf during such absence. Provided, however, if a Participant is on an Authorized Leave of Absence and is receiving worker's compensation during such Authorized Leave of Absence, and if the Employer so elects in the Joinder Agreement, such Participant shall be credited with Service for such period for purposes of vesting only (and not for purposes of benefits) but no Employee contributions shall be made with respect to the Participant for such period.

- (4) The expiration of the term of office of an elected official shall not be considered as interrupting continuity of employment, provided the official is re-elected for a consecutive term.
- (5) Any reference in this Plan to the number of years of service of an Employee shall include fractional portions of a year.
- (6) With respect to a Participant who was previously 100% vested in any other Municipality's qualified retirement plan prior to becoming a Participant in this Plan, such Participant's "Service" for purposes of determining years of service for vesting under this Plan shall include the Participant's last continuous period during which the Participant was an employee of the other Municipality.

Credit for service with the Employer shall not be granted for any period subsequent to the Effective Date during which the Employee did not participate in the Plan and Employee contributions to the Plan and Fund were not made by or for him except as specified above.

- (II) Severance Benefit: The pension benefit described in Section 6.1 herein.
- (mm) Spouse: Effective as of June 26, 2013, for Federal tax purposes which may apply to qualified retirement plans under Code Section 401(a), the terms "spouse," "husband," and "wife" include an individual married to a person of the same sex if the individuals are lawfully married under state law, and the term "marriage" includes such marriage between individuals of the same sex, and a marriage of same-sex individuals that was validly entered into in a state whose laws authorize the marriage of two individuals of the same sex even if the married couple is domiciled in a state that does not recognize the validity of same-sex marriages. For all other Plan purposes and which are not required for Federal tax purposes as described in the preceding sentence, the term "spouse" will be defined as a spouse which is legally recognized in the State of Oklahoma.
- (nn) Trust Service Provider: The person appointed by the Trustees to supervise operation of the Oklahoma Municipal Retirement Fund and to assist participating Municipalities in the adoption and operation of the Plan.
- (00) Trustee: The Trustees appointed pursuant to the Trust Indenture establishing the Oklahoma Municipal Retirement Fund.

- (pp) U. S. Consumer Price Index: The Consumer Price index for all items as reported in the Monthly Labor Review for the month of December of the immediately preceding calendar year as published by the United States Department of Labor.
- (qq) Year of Service: A 12 consecutive month period of service commencing on the Employee's Employment Commencement Date, and any anniversary thereof.
- 2.2 Construction: The masculine gender, where appearing in the Plan, shall be deemed to include the feminine gender, unless the context clearly indicates to the contrary. The words "hereof," "herein," "hereafter" and other similar compounds of the word "here" shall mean and refer to the entire Plan, not to any particular provision or Section.

<u>ARTICLE III.</u>

Contributions

- 3.1 Eligibility: An Employee, as defined in the Joinder Agreement, who has satisfied all the requirements set forth in the Joinder Agreement shall be eligible to participate in the Plan. Any person who has been classified by the Employer as an independent contractor and has had his compensation reported to the Internal Revenue Service on Form 1099 but who has been reclassified as an "employee" (other than by the Employer) shall not be considered as an eligible Employee who can participate under this Plan; provided, if the Employer does reclassify such worker as an "Employee," for purposes of this Plan, such reclassification shall only be prospective from the date that the Employee is notified by the Employer of such reclassification.
- 3.2 Contributions by Employer: The Employer shall make contributions to the Fund in such amounts and at such times as the City Council shall determine, acting under the advice of the Plan's actuarial firm. All contributions made by the Employer to the Fund shall be irrevocable, and shall be used for the exclusive benefit of the Participants covered by the Plan to pay benefits under the Plan, or to pay expenses of the Fund. Forfeitures arising because of death prior to Retirement, severance of Employment before a Participant becomes eligible for a Pension, or any other reason shall be applied to reduce the cost of the Plan, not to increase the benefits otherwise payable to the Participants.
- 3.3 Contributions by Participants: Unless the Employer selects the Non-Contributory Option, each Participant shall contribute to the cost of providing benefits under this Plan while he remains a Participant. Such annual contributions shall be the product of (i) the percentage selected in Section 8 of the Joinder Agreement, and (ii) the Participant's Compensation.

Any required contributions by Participants shall be made by payroll deductions for each pay period, or any series of pay periods as the Employer may deem most convenient, during the full time of employment as an Employee. The City Council may, however, approve payment of such contributions in a manner other than payroll deductions in any specific case or cases. (In any event a Participant shall be deemed to consent and agree to the payroll deductions as provided for herein.) If a Participant is granted a non-paid leave of absence authorized for any reason, his continued participation in the Plan will depend upon his contributions being continued at the same rate and made by or for him during such absence. While such contributions are continued during such non-paid leave of absence, the Participant's Compensation shall be deemed to have continued at the same rate for the purpose of computing the Participant's Average Monthly Compensation.

Provided, however, if a Participant is on an Authorized Leave of Absence and is receiving worker's compensation during such Authorized Leave of Absence, such Participant shall be credited with Service for such period for purposes of vesting only and not for purposes of benefits if the Employer so elects in Section 7G of the Joinder Agreement, but no Participant contributions shall be made with respect to the Participant for such period.

Each Employee in the service of the Employer on the Effective Date of the Plan hereof may become a Participant in the Plan when first eligible by signing a written notice of participation, agreeing to be bound by the terms and conditions hereof, and authorizing the Employer to deduct from his Compensation the contributions required of him as provided in the Joinder Agreement,

hereof, and he shall be subjected to all other provisions of the Plan beginning on such date. An Employee in the Service of the Employer who does not so elect to become a Participant as of the date he is first eligible to do so, may so elect to become a Participant as of the first day of his pay period coincidental with or next following the date as of which he gives to the Employer written authority to commence deductions from his Compensation for the contributions required of him as is provided in the Joinder Agreement. However, the Service of such a Participant shall not include the period of such voluntary nonparticipation prior to the date as of which the Employee elects to become a Participant in the Plan.

Each Employee employed on or after the original Effective Date of the Plan hereof shall, as a condition of employment, become a Participant in the Plan as of the date on which he is first eligible by signing a written notice of participation agreeing to be bound by the terms and conditions hereof, and authorizing the Employer to deduct from his Compensation any contributions required of him as provided in the Joinder Agreement hereof, and he shall be subject to all other provisions of the Plan beginning on such date.

For each Employee who becomes a Participant in this Plan on the original Effective Date of the Plan, Participant contributions and his participation shall first begin for the pay period commencing on, or next following, that date. For each Employee who becomes a Participant in this Plan after the original Effective Date of the Plan, Participant contributions and his participation shall begin for the pay period commencing on or next following the date he becomes a Participant.

Such Participant contributions shall be fully vested in the contributor Participant at all times. Upon retirement, death or termination of employment of a Participant for any reason, the retired or terminated Employee, or his Beneficiary as the case may be, shall have the option to receive, in lieu of any and all other benefits provided herein, his Contribution Accumulation. Furthermore, the value of the total benefits payable to the Participant and/or his Beneficiary shall in no event be less than his Contribution Accumulation as of the time of his termination of employment. However if any benefit of any other kind is paid under this Plan, to or on behalf of a Participant, no Contribution Accumulation shall be paid, but shall be deemed to have been included in the value of the benefit so paid, unless the total value of such other benefit payments finally paid shall be less than such Contribution Accumulation as of the time of the Participant's termination of employment, in which case the difference shall be paid in a lump sum to the Participant and/or his Beneficiary.

3.4 Pick-up Contributions: If the Employer elects the Pick-Up Option in the Joinder Agreement, all Participants shall be required to make the contributions specified in the Joinder Agreement. These contributions shall be picked up and assumed by the Employer and paid to the Fund in lieu of contributions by the Participant. Such contributions shall be designated as Employer contributions for federal income tax purposes. Each Participant's Compensation will be reduced by the amount paid to the Fund by the Employer in lieu of the required contribution by the Participant. These contributions shall be excluded from the Participant's gross income for federal income tax purposes and from wages for purposes of withholding under Sections 3401 through 3404 of the Code in the taxable year in which contributed. No Participant shall have the option of receiving the contributed amounts directly as compensation. Contributions made by the Employer under this election shall be designated as Participant contributions for purposes of vesting, and determining Participant rights and the Participant's Contribution Accumulation and

shall be allocated to a separate account. A private letter ruling is required if the sponsor/employer wishes a ruling on pick-up contributions.

3.5 **Transfer of Contributions:** All Employer and Participant contributions shall be directly or immediately allocated, paid or delivered to the City Treasurer, as Treasurer of the Plan. Such contributions shall be transferred and transmitted by the City Treasurer to the Fund for credit as soon as administratively feasible.

ARTICLE IV.

Requirements for Retirement Benefits

- Normal Pension: A Participant shall be eligible for a Normal Pension if his employment is terminated on or after his Normal Retirement Date, or if his employment classification has changed such that he is no longer eligible to participate in this Plan on or after his Normal Retirement Date, provided he has met the 100% vesting requirements. Payment of a Normal Pension shall commence as of the first day of the month coinciding with or next following Retirement or change in employment classification, as applicable, and the last payment shall be made as of the first day of the month in which the death of such Participant occurs; provided however, that at the time of his death, if such Participant has received less than the number of monthly payments elected by the Employer in Section 7 of the Joinder Agreement, his Pension payments shall continue to his Beneficiary or Beneficiaries until a total of such number of monthly payments as elected have been made to such Participant and such Beneficiary or Beneficiaries. Normal Pension payments shall not be suspended for a retired Participant who returns to work for the Employer in an employment classification which is not eligible to participate in this Plan.
- 4.2 Early Pension: A Participant may elect early Retirement and be eligible for an Early Pension if his employment is terminated on or after his 55th birthday and before his Normal Retirement Date, provided he has met the 100% vesting requirements. Payment of an Early Pension shall commence as of the Participant's Normal Retirement Date. However, if a Participant requests the Committee to authorize the commencement of his Early Pension as of the first day of any subsequent month which precedes his Normal Retirement Date, his Pension shall commence as of the beginning of the month so requested, but the amount thereof shall be reduced as provided in Section 5.2. The last payment of an Early Pension shall be made as of the first day of the month in which the death of the retired Participant occurs; provided however, that if the retired Participant has received less than the monthly payments as elected in Section 7 of the Joinder Agreement at the time of his death, his Pension payments shall continue to his Beneficiary or Beneficiaries until a total of such monthly payments have been made to such Employee and such Beneficiary or Beneficiaries.
- 4.3 **Disability Pension:** A Participant shall be eligible for a Disability Pension if his employment is terminated by reason of Disability, before his Normal Retirement Age, provided he has met the 100% vesting requirements. Payment of a Disability Pension shall commence as of the first day of the month coincidental with or next following the date of Retirement. The last payment shall be made as of the first day of the month in which the death of the retired Employee occurs, or if Disability ceases prior to his Normal Retirement Date, the first day of the month in which Disability ceases.

Disability under the Plan shall be considered total and permanent, if on the basis of a medical examination by a doctor or clinic appointed by the Committee, the Committee finds that the Participant has a physical or mental condition which totally and presumably permanently prevents him from engaging in any substantial gainful employment with the Employer.

Notwithstanding any other provisions of this Section, no Participant shall qualify for a Disability Pension if the Committee determines that his Disability results from (a) chronic alcoholism, (b) addition to narcotics, (c) an injury suffered while engaged in a felonious or criminal act or enterprise, or (d) service in the armed forces of the United states which entitles the Employee to a veteran's disability pension.

Disability shall be considered to have ended and a Disability Pension shall cease if, prior to his Normal Retirement Age, the Participant (a) engages in any substantial gainful employment except for such employment as is found by the Committee to be for the primary purpose of rehabilitation or not incompatible with a finding of total and permanent Disability, or (b) has sufficiently recovered, in the opinion of the Committee based on a medical examination by a doctor or clinic appointed by the Committee to be able to engage in regular employment with the Employer and refuses an offer of employment by the Employer, or (c) refuses to undergo any medical examination requested by the Committee provided that a medical examination shall not be required more frequently than twice in any calendar year.

If Disability ceases before a retired Participant attains his Normal Retirement Date and the Participant is re-employed by the Employer, the Pension payable upon his subsequent Retirement shall be determined in accordance with the provisions of Section 10.10.

- 4.4 **Deferred Vested Pension:** A Participant shall be eligible for a Deferred Vested Pension, if his employment is terminated before his 55th birthday and after he has met the 100% vesting requirements. Payment of a Deferred Vested Pension shall commence as of the Participant's Normal Retirement Date. However, if the Participant requests the Committee to authorize the commencement of his Deferred Vested Pension as of the first day of the month coinciding with or next following his 55th birthday, or as of the first day of any subsequent month which precedes his Normal Retirement Date, his Pension shall commence as of the first day of the month so requested, but the amount thereof shall be reduced as provided in Section 5.4.
- 4.5 **Pensions for Former Employees:** If a Participant's Service with the Employer terminates, but his Service continues by virtue of his employment with a Municipality other than the Employer, he, his spouse or other Beneficiaries shall only be then, or later become entitled to, and limited to, such rights, benefits and options of any kind, under this Plan, if any, in the amounts and on the terms and conditions, as provided in Article VIII-Employment Transfers.

4.6 Termination of Employment - Vesting of Benefits:

- (a) General: Except as provided in Sections 10.12 and 10.13 hereof, when a Participant ceases to be a Participant for any reason, he shall have vested and nonforfeitable rights in his Accrued Benefits as set forth in one of the following vesting schedules as may be elected by the Employer in the Joinder Agreement:
- (i) Ten Year Cliff Vesting Schedule. The Ten Year Cliff Vesting Schedule is as follows:

Percent of Accrued

Years of Service

Benefit Vested

Less than: 10 0%

At least: 10 100%

(ii) Seven Year Cliff Vesting Schedule. The Seven Year Cliff Vesting Schedule is as follows:

Years of Service
Percent of Accrued
Benefit Vested

Less than: 7
At least: 7
0%
100%

(iii) Five Year Cliff Vesting Schedule. The Five Year Cliff Vesting Schedule is as follows:

Percent of Accrued

Years of Service

Benefit Vested

Less than: 5
At least: 5

O%
100%

ARTICLE V.

Amount of Retirement Benefits

5.1 Normal Pension:

- (a) Basic Formula: A Participant who meets the requirements for a Normal Pension shall receive a monthly amount equal to the product of (1), (2), and (3) as follows:
- (1) The percentage associated with the Plan Option elected by the Participant in Section 7B of the Joinder Agreement; <u>multiplied</u> by
 - (2) His Average Monthly Compensation; and multiplied by
- (3) The number of his Years of Service credited with the Employer (but not with any other Municipality), subject to the limitations in Section 7B of the Joinder Agreement.
- Cost-of-Living Adjustment: If the Cost-of-Living Option is elected in the **(b)** Joinder Agreement, the monthly amount of Normal Pension determined above under Subsection (a) of this Section, or the amount of any optional form of Pension payable in lieu thereof to a retired Participant or his contingent Beneficiary, shall be increased or decreased annually while payable, commencing with the payment due on the first day of July coinciding with or next following the later of (1) the effective date of the Cost-of-Living option, or (2) the date of the Participant's Retirement, and continuing thereafter on the first day of each subsequent July during which the Pension is payable. Each such increase or decrease shall be related to a change in the cost-of-living based on the percentage change, if any, determined by a comparison of the U. S. Consumer Price Index (as defined in Section 2.1(pp)) for the December next preceding the July of the determination, with such U. S. Consumer Price Index for the December one year earlier; provided however, that such yearly increase or decrease, if any, shall be limited to a maximum change of three percent (3%); and provided further, that such yearly decrease, if any, shall not reduce the amount of Pension so adjusted, below the level established at the time of Retirement.

5.2 Early Pension:

- (a) Basic Formula: A Participant who meets the requirements for an Early Pension shall receive a monthly amount which shall be computed in the same manner as a Normal Pension, considering his Compensation and Service credited with the Employer prior to Retirement. If payment of an Early Pension commences prior to the Participant's Normal Retirement Date, the amount determined above shall be reduced by 5% each full year plus 5% pro-rata for the number of months in the period between the date as of which the Pension begins and the Normal Retirement Date.
- (b) Cost-of-Living Adjustment: If the Cost-of-Living option is elected in the Joinder Agreement, the monthly amount of Early Pension determined above under Subsection (a) of this Section, or the amount of any optional form of Pension payable in lieu thereof to a retired Participant or his contingent Beneficiary, shall be subject to annual cost-of-living adjustments in the same manner as provided in Subsection 5.1(b), except that no such adjustment

shall be applicable for any period before the Early Pension payments commence, and for this purpose only, the date such payments commence shall be treated as the Participant's Retirement Date.

5.3 Disability Pension:

- (a) Basic Formula: A Participant who meets the requirements for a Disability Pension shall receive a monthly amount which shall be computed in the same manner as a Normal Pension, considering his Compensation and Service credited with the Employer prior to Retirement.
- (b) Cost-of-Living Adjustment: If the Cost-of Living Option is elected in the Joinder Agreement, the monthly amount of Disability Pension determined above under Subsection (a) of this Section shall be subject to a cost-of-living adjustment in the same manner as provided in Subsection 5.1(b), except that such adjustment shall not be applicable for any period before the Disability Pension payments commence, and for this purpose only, the date such payments commence shall be treated as the Participant's Retirement Date.

5.4 Deferred Vested Pension:

- (a) Basic Formula: A Participant who meets the requirements for a Deferred Vested Pension shall receive a monthly amount which shall be computed in the same manner as a Normal Pension, considering his Compensation and Service credited with the Employer prior to the termination of his employment. If payment of a Deferred Vested Pension commences prior to the Participant's Normal Retirement Date, the amount determined above shall be reduced by 5% each full year plus 5% pro-rata for the number of months in the period between the date as of which the Pension begins and Normal Retirement Date.
- (b) Cost-of-Living Adjustment: If the Cost-of-Living Option is elected in the Joinder Agreement, the monthly amount of Deferred Vested Pension determined above under Subsection (a) of this Section shall be subject to a cost-of-living adjustment in the same manner as provided in Subsection 5.1(b), except, that such adjustment shall not be applicable for any period before the Deferred Vested Pension payments commence, and for this purpose only, the date such payments commence shall be treated as the Participant's Retirement Date.
- 5.5 Accrued Credits and Vested Benefits Preserved: The adoption of a new Joinder Agreement by an Employer shall not operate to exclude, diminish, limit, or restrict the amount, payments or continuation of payments of benefits accrued up to the Effective Date of the most recent Joinder Agreement. The amount of such Accrued Pension benefits, if any, in the course of payment immediately prior to such date, shall be continued under the provisions of such Previous Plan, in the same manner and amounts, subject to the provisions of the Retiree Plan Improvement Option in Section 10 of the Joinder Agreement.

ARTICLE VI.

Severance and Death Benefits

6.1 Severance Benefit: Upon termination of a Participant's employment with the Employer prior to his Retirement, his contributions to the Fund shall cease and he shall be entitled to receive a Severance Benefit equal in amount to the Participant's Contribution Accumulation, which shall be in lieu of all other benefits under this Plan. Payment of such benefit shall be made in a lump sum as soon as administratively feasible after the date of termination of the Participant's employment and the Participant's request for payment. If such termination of employment occurs after the Participant has met the requirements for Deferred Vested Pension, he may receive in lieu of such Contribution Accumulation, the Pension for which he is eligible under the provisions of Article IV, unless the Participant elects to receive such Contribution Accumulation.

If any benefit of any other kind is paid under this Plan to or on behalf of a Participant, no Severance Benefit shall be paid, but shall be deemed to have been included in the value of the other benefit, unless the total of such other benefit payments finally made shall be less than his Contribution Accumulation at the time of the Participant's termination of employment, in which case the difference shall be paid to the terminated Participant if living, or if deceased, to his Beneficiary.

- 6.2 Death Prior to Commencement of Pension: Upon the death of an active Participant or a retired Participant prior to the earlier of the date the Committee approves the commencement of his pension payments or the date fixed for commencement of his Pension payments, the Beneficiary designated by the Participant or retired Participant shall be paid a Death Benefit in the form of a Pension unless a spouse's pension becomes payable under Section 6.4.
- Subject to the further provisions of Subsection (b) of this Basic Formula: Section 6.2, the amount of the Death Benefit Pension shall be equal to fifty percent (50%) of the monthly amount of the Normal Pension, as determined in Section 5.1 which the deceased Participant had accrued at the time of his death considering the Participant's Compensation and Service with the Employer prior to the date of his death. Payment of the Death Benefit Pension under this Subsection shall commence as of the first day of the month coincident with or next following the Employee's death. The last payment shall be made upon completion of the number of monthly payments in the aggregate as elected in Section 7A of the Joinder Agreement. Notwithstanding the foregoing, if any spouse's Pension becomes payable under Section 6.4, or if any optional Pension was elected by such a retired Participant, and becomes effective under Article VII, no such Death Benefit Pension under this Section 6.2 shall be paid at that time. The terms of such Spouse's Pension or optional Pension as the case may be, shall control payments after such death and the Death Benefit Pension provided for under this Section shall be thereby canceled or inapplicable, except that upon the death of the spouse or the contingent Beneficiary, his estate shall be paid the excess, if any, of the Participant's Contribution Accumulation as of the date the Pension commenced or as of the date of death, if earlier, over the sum of the benefit payments other than payments derived from Disability, previously received by the spouse or the contingent Beneficiary.

(b) Cost-of-Living Adjustment: If the Cost-of-Living Option is elected in the Joinder Agreement, the monthly amount of Death Benefit Pension determined above under Subsection (a) of this Section 6.2 or the amount of any optional form of Pension payable in lieu thereof to the Beneficiary entitled thereto, shall be subject to a cost-of-living adjustment in the same manner as provided in Subsection 5.1(b).

6.3 Death After Commencement of Pension:

- (a) Normal Pension, Early Pension or Deferred Vested Pension: Upon the death of a retired Participant after the earlier of the date the Committee approves the commencement of his Normal Pension payments, Early Pension payments or Deferred Vested Pension payments as the case may be, or the date fixed for commencement of his Normal Pension payments or Early Pension payments, Deferred Vested Pension payments as the case may be, and before he has received the number of monthly payments elected in Section 7A of the Joinder Agreement, his Beneficiary shall be entitled to a Death Benefit to be provided in the form of a Pension.
 - (1) Basic Formula: Subject to the further provisions of Subsection (a)(2) of this Subsection 6.3(a), the amount of such Death Benefit Pension shall be equal to the monthly amount of Normal Pension, Early Pension or Deferred Vested Pension which the deceased retired Participant was eligible for or receiving at the time of his death. Payment of such Death Benefit Pension shall commence as of the first day of the month coincident with or next following the retired Participant's date of death. The last payment shall be made upon the completion of the number of monthly payments in the aggregate as elected in Section 7A of the Joinder Agreement to the retired Participant and the Beneficiary, if living, or if deceased, the estate of the Beneficiary. This Death Benefit Pension shall not be in addition to, but shall be one and the same as the continuation of Pension as provided in Section 4.1, Section 4.2, or Section 4.4, as the case may be. However, no such Death Benefit shall be paid but shall be canceled and inapplicable, if an optional form of payment is elected and becomes effective under Article VII hereof.
 - (2) Cost-of-Living Adjustment: If the Cost-of-Living Option is elected in the Joinder Agreement, the monthly amount of Death Benefit Pension determined above under Subsection (a)(1) of this Section 6.3 or the amount of any optional form of Pension payable in lieu thereof to the Beneficiary or estate entitled thereto, shall be subject to a cost-of-living adjustment in the same manner as provided in Subsection 5.1(b).
- (b) Disability Pension: Upon the death of a retired Participant who was unmarried or legally separated from his or her spouse, after the earlier of the date the Committee approves the commencement of his Disability Pension payments, or the date fixed for commencement of his Disability Pension payments and before he has received a total of the number of monthly Pension payments as elected in Section 7A of the Joinder Agreement, and before the cessation of his Disability if such death occurs prior to his Normal Retirement Date, his Beneficiary shall be entitled to a Death Benefit to be provided in the form of a Pension.
 - (1) Basic Formula: Subject to the further provision of Subsection (b)(2) of this Section 6.3(b), the amount of such Death Benefit Pension shall be equal to the monthly amount of Disability Pension which the deceased retired Participant was eligible for or

receiving at the time of his death. Payment of such Death Benefit Pension shall commence as of the first day of the month coincident with or next following the retired Participant's date of death. The last payment shall be made upon the completion of the number of monthly payments in the aggregate as elected in Section 7A of the Joinder Agreement to the retired Participant and the Beneficiary, if living, or if deceased, to the estate of the Beneficiary.

If the death of such retired Participant occurs after the cessation of his Disability and before his Normal Retirement Date, and the total Disability Pension payments he had received was less than his Contribution Accumulation as of the date of commencement of payments of such Disability Pension, or as of the date of his death, if earlier, then his Beneficiary shall be entitled to a Death Benefit. The amount of such Death Benefit shall be the excess of the retired Participant's said Contribution Accumulation over the sum of such Pension payments, if any, previously received by the retired Participant. Such Death Benefit shall be paid in cash in a single sum within 30 days after the date of death.

If the retired Participant was married and not legally separated from his or her spouse at the time of death, the applicable Death Benefit shall be that as provided in Section 6.4(b).

(2) Cost-of-Living Adjustment: If the Cost-of-Living Option is elected in the Joinder Agreement, the monthly amount of Death Benefit Pension determined above under Subsection (b)(1) of this Section 6.3 shall be subject to a cost-of-living adjustment in the same manner as provided in Section 5.1(b).

6.4 Spouse's Pension:

(a) In-Service Death:

- (1) Eligibility Requirements: The surviving spouse of a deceased Participant shall be eligible for a Spouse's Pension with payments commencing on the first day of the month coinciding with or next following the Participant's date of death and payable for the spouse's lifetime, or until the spouse's remarriage, provided that the Participant, as of the date of his or her death, (a) was continuing in the active Service of the Employer, (b) had met the 100% vesting requirement, (c) had not retired or begun receiving his or her Normal Pension, and (d) was not legally separated from the surviving spouse.
- (2) Amount of Spouse's Pension: A surviving spouse who meets the eligibility requirements under Subsection (a)(1) of Section 6.4 above shall receive a monthly amount of Spouse's Pension equal to fifty percent (50%) of the amount determined in Section 5.1 for a Normal Pension considering the Participant's Compensation and Service with the Employer to the date of his death. However, if the surviving spouse is more than ten years younger than the retired Participant on the date of his death, the Spouse's Pension payable under this Subsection shall be reduced by one percent (1%) for each such year of age difference in excess of ten (10) years to compensate for the longer period of expected payments.
- (3) Cost-of-Living Adjustment: If the Cost-of-Living Option is elected in the Joinder Agreement, the monthly amount of Spouse's Pension determined above under

Subsection (a)(2) of this Section 6.4 shall be subject to a cost-of-living adjustment in the same manner as provided in Subsection 5.1(b).

- (b) Post-Disability Retirement Death: The surviving spouse of a deceased, retired Participant, who was receiving or was entitled to receive a Disability Pension on the date of his or her death and who had received less than the number of monthly payments as elected in Section 7A of the Joinder Agreement of such Disability Pension, shall be eligible for a Spouse's Pension.
 - (1) Basic Formula: Subject to the further provisions of Subsection (b)(2) of this Section 6.4, the amount of such Spouse's Pension shall be equal to the monthly amount of Disability Pension which the deceased retired Participant was eligible for or receiving at the time of his death. Payment of such Spouse's Pension shall commence as of the first day of the month coincident with or next following the retired Participant's date of death. The last payment shall be made upon the completion of the number of monthly payments in the aggregate as elected in Section 7A of the Joinder Agreement to the retired Participant and the surviving spouse, or if the surviving spouse dies before such completion of payments, the remaining payments shall be made to the estate of the deceased spouse.
 - (2) Cost-of-Living Adjustment: If the Cost-of-Living Option is elected in the Joinder Agreement, the monthly amount of Spouse's Pension determined above under Subsection (b)(1) of this Section 6.4 shall be subject to a cost-of-living adjustment in the same manner as provided in Section 5.1(b).
- (c) Post-Termination Death Benefit After Deferred Vested Termination: The surviving spouse of a deceased, retired Participant, who was entitled to receive a Deferred Vested Pension but had not yet received any payments on the date of his or her death, shall be eligible for a Spouse's Pension. Such Pension will commence on the first day of the month coinciding with or next following the later of: (i) the Participant's date of death or (ii) the earliest date of which the Participant could have begun receiving payments in accordance with Section 5.4. The last payment shall be made upon the death or remarriage of the surviving spouse. A final death benefit is the excess, if any, of the Participant's Contribution Accumulation over the sum of the payments made to the Spouse.
 - (1) Basic Formula: Subject to the further provisions of Subsection (c)(2) of this Section 6.4, the amount of such Spouse's Pension shall be equal to fifty percent (50%) of the Deferred Vested Pension to which the deceased, retired Participant was entitled to receive commencing on his or her Normal Retirement Date and reduced in accordance with Section 5.4 for the period between the date the Pension begins and the Normal Retirement Date. However, if the surviving spouse is more than ten (10) years younger than the retired Participant on the date of his death, the Spouse's Pension payable under this Subsection shall be reduced by one percent (1%) for each such year of age difference in the excess of ten (10) years to compensate for the longer period of expected payments.
 - (2) Cost-of-Living Adjustment: If the Cost-of-Living Option is elected in the Joinder Agreement, the monthly amount of spouse's Pension determined above under

Subsection (c)(1) of this Section 6.4 shall be subject to a cost-of-living adjustment in the same manner as provided in Subsection 5.1(b).

Beneficiary or Beneficiary: Each active or retired Participant may designate a primary Beneficiary or Beneficiaries and, in addition, may name a contingent Beneficiary or Beneficiaries to receive any benefit that may become payable under Article VI hereunder by reason of his death. If a Participant designates more than one Beneficiary, each shall share equally unless the Participant specifies a different allocation or preference. Such designation shall be made upon forms furnished by the Employer and may be revoked or changed at any time and from time to time without notice to any Beneficiary, and shall not be effective unless and until filed with the Committee. Further, the written designation of the Participant's spouse may be voided upon divorce of the Participant if required by applicable state law. If a Participant fails to designate a Beneficiary, or if no designated Beneficiary survives the Participant, the Death Benefit shall be paid to the Participant's spouse, if living, or otherwise, to the estate of the Participant. Neither the Employer, the Board of Trustees, nor the Fund shall be named as a Beneficiary.

For the purpose of this Plan, the production of a certified copy of the death certificate of any Employee or other person shall be sufficient evidence of death, and the Committee shall be fully protected in relying thereon. In the absence of such proof, the Committee may rely upon such other evidence of death as it deems necessary or advisable.

- 6.6 Severance of Death Benefits for Former Employees: If a Participant's Service with the Employer terminates, but his Service continues by virtue of his employment with a Municipality other than the Employer, he, his spouse or other Beneficiaries shall only be then, or later become, entitled to and limited to such rights, benefits and options of any kind, under this, if any, in the amounts and on the terms and conditions, as provided in Article VIII, Employment Transfers.
- 6.7 HEART Act Provision: In the case of a death occurring on or after January 1, 2007, if a Participant dies while performing qualified military service (as defined in Code §414(u)), the survivors of the Participant are entitled to any additional benefits (other than benefit accruals relating to the period of qualified military service, but including vesting service credit for such period and any ancillary life insurance or other survivor benefits) that would have been provided under the Plan had the Participant resumed employment on the day preceding the participant's death and then terminated employment on account of death.

ARTICLE VII.

Optional Retirement Benefits

- 7.1 Joint and Survivor Options: By filing an application with the Authorized Agent prior to the start of payments, a married Participant not legally separated from his or her spouse may designate such spouse as his contingent pensioner and elect to receive a Pension payable in accordance with one of the following Actuarially Equivalent options in lieu of the Pension to which he may otherwise become entitled upon Retirement.
- Option A Joint and 100% Survivor Annuity. A reduced Pension payable monthly during the lifetime of the Participant with the provision that 100% of such monthly benefit shall be payable to the Participant's contingent pensioner in monthly installments commencing on the first day of the month following the month in which the Participant died and continuing thereafter during the remaining lifetime of such contingent pensioner through the last monthly payment on or prior to the contingent pensioner's death. The reduced Pension payable to the Participant shall be eighty-three percent (83%) plus or minus one percent (1%) for each year to the nearest year that the contingent pensioner is older or younger respectively than the Participant multiplied by the Pension payable to the Participant in the normal form.
- Option B Joint and 50% Survivor Annuity. An adjusted Pension payable monthly during the lifetime of the Participant with the provision that fifty percent (50%) of such monthly benefit shall be payable to such Participant's contingent pensioner in monthly installments commencing on the first day of the month following the month in which the Participant dies and continuing thereafter during the remaining lifetime of the contingent pensioner through the last monthly payment on or prior to such contingent pensioner's death. The reduced Pension payable to the Participant shall be ninety-two percent (92%) plus or minus one-half of one percent (0.5%) for each year to the nearest year that his contingent pensioner is older or younger respectively than the Participant multiplied by the Pension payable to the Participant in the normal form.
- Option C Joint and 66 2/3% Survivor Annuity. An adjusted Pension payable for the joint lifetime of the Participant and his contingent pensioner, and upon the death of either, payments in the amount of sixty-six and two-thirds percent (66 2/3%) of such adjusted Pension shall be continued to the contingent pensioner during the contingent pensioner's lifetime through the last monthly payment on or prior to such contingent pensioner's death. The reduced Pension payable to the Participant shall be ninety-three percent (93%) plus or minus seven-tenths of one percent (0.7%) for each year to the nearest year that the contingent pensioner is older or younger respectively than the Participant multiplied by the Pension payable to the Participant in the normal form.
- 7.2 Other Forms of Payment: If the Employer has elected in the Joinder Agreement to provide additional optional benefit forms, the Committee may, in its sole discretion, at the request of an Participant (or contingent pensioner), direct that any benefit provided by the Plan be paid in one of the following forms, provided that payments to the Participant (or contingent pensioner) have not yet commenced and that payments in such other form shall be the Actuarial Equivalent of the benefit otherwise payable. The optional forms of payment are as follows:

- Option D Insured Annuity. Under this form, the payee will receive a nontransferable annuity purchased from a duly licensed insurance company under either an individual or group annuity contract. Such annuity may be in any of the forms otherwise payable hereunder.
- Option E Periodic Installments. Under this form, the payee will receive periodic installments over a period of years not to exceed life expectancy or the life expectancy of the payee and his designated Beneficiary. If his death occurs after payments commenced, any remaining installments will be paid to his designated Beneficiary, or Beneficiaries, either periodically over the remainder of the period originally established for the payee or in a lump sum, as selected by the Committee. No future Cost-of Living adjustments will be made or considered in calculating the payment under this optional form.
- Option F Lump-Sum Payment. Under this form, the payee will receive a single sum payment in cash. No future Cost-of-Living adjustments will be made or considered in calculating the payment under this optional form.
- **Option G** Combination. Under this form, the payee will receive a combination of Option F and Option F or Option F, as selected by the payee.

The calculation of amounts payable under Option E and Option F above shall be based on actuarial tables contained in Appendix I. The underlying interest rate shall be seven and one-half percent (7½%).

Annuity contracts purchased under Option D above must be priced on a basis deemed not to be discriminatory under Title VII of the Civil Rights Act.

The Committee shall, if it deems appropriate, require a Participant (or contingent pensioner) to submit evidence of good health as a condition to receipt of any such form of payment, particularly any lump sum payment.

If a Pension payable under this Plan is less than fifty dollars (\$50.00) per month, the Committee may direct that, in lieu of such Pension, the Actuarial Equivalent thereof shall be paid in a lump sum, or in a series of uniform monthly, quarterly, or annual amounts for life or for a designated period of time.

For any distribution notice issued in Plan Years beginning after December 31, 2006, any reference to the 90-day maximum notice period prior to distribution in applying the notice requirements of Code §§402(f) (the rollover notice), or 411(a)(11) (Participant's consent to distribution) will become 180 days.

7.3 Restrictions on Optional Forms: If payments have not yet commenced to a Participant, a Participant may elect, change, or revoke an option if his election, change, or revocation is filed in writing with the Authorized Agent. However, an election to receive benefits in one of the forms described in Section 7.2 requires Committee approval. In the event a Participant dies after he has begun to receive benefits under this Plan, his beneficiary or contingent pensioner shall not be entitled to change the form of payment of the benefit. A Participant receiving a Disability Pension is not eligible for any of the options. Notwithstanding anything in this Section to the contrary, a Participant may elect an option without the approval of the Authorized Agent at any

time within the six month period next following the adoption of this Plan by the Employer, provided that the Participant is in the employ of the Employer at the time the election is made.

An election made pursuant to this Article shall become inoperative in the event that no contingent pensioner is surviving upon the Participant's Retirement Date.

- If a Participant who makes an election pursuant to the requirements of this Section continues in the Employer's employ after his Normal Retirement Date, no Pension payments shall be made during the period of continued employment. If the Participant dies during such continued employment and the contingent pensioner survives him, the election shall become operative so that the contingent pensioner shall receive a Pension in accordance with the option elected commencing on the first day of the month coinciding with or next following the death of the Participant. In the event the contingent pensioner predeceases the Participant during such continued employment, the election shall not become operative.
- 7.4 Other Benefits Canceled by Option: Any Contribution Accumulation, Pension, Severance, Death, or other benefit that would otherwise have become payable under this Plan shall be canceled and superseded by an option elected under Section 7.1 or any other form of payment elected under Section 7.2 as of the date such option or other form of payment commences.
- 7.5 **Options by Former Employee:** The provisions of this Article VII shall be applicable to any former Employees entitled thereto under the provisions of Article VIII-Employment Transfers.
- 7.6 Rollover to Another Plan or IRA: Notwithstanding any provision of the Plan to the contrary that would otherwise limit a Distributee's election under this Section, a Distributee may elect, at the time and in the manner prescribed by the Committee, to have any portion of an Eligible Rollover Distribution paid directly to an Eligible Retirement Plan specified by the Distributee in a Direct Rollover. The Committee shall establish procedures for implementing such Direct Rollover distribution.
- (a) **Definitions:** For purposes of this Section 7.6, the following definitions shall apply:
- (i) "Eligible Rollover Distribution": An "Eligible Rollover Distribution" is any distribution of all or any portion of the balance to the credit of the Distributee, except that an Eligible Rollover Distribution does not include: any distribution that is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the Distributee or the joint lives (or joint life expectancies) of the Distributee and the Distributee's designated Beneficiary, or for a specified period of ten years or more; any distribution to the extent such distribution is required under Section 401(a)(9) of the Code; and the portion of any distribution that is not includable in gross income. With respect to distributions made after December 31, 2001, a portion of a distribution shall not fail to be an eligible rollover distribution merely because the portion consists of after-tax employee contributions which are not includible in gross income. However, such portion may be paid only to an individual retirement account or annuity described in Section 408(a) or (b) of the Code, or to a qualified defined contribution plan described in Section 401(a) or 403(a) of the Code that agrees to separately account for amounts so

transferred, including separately accounting for the portion of such distribution which is includible in gross income and the portion of such distribution which is not so includible.

- "Eligible Retirement Plan": An "Eligible Retirement Plan" is an individual retirement account described in Section 408(a) of the Code, an individual retirement annuity described in Section 408(b) of the Code, an annuity plan described in Section 403(a) of the Code, a qualified trust described in Section 401(a) of the Code, an annuity contract described in Section 403(b) of the Code, an eligible plan under Section 457(b) of the Code which is maintained by a state, political subdivision of a state, or any agency or instrumentality of a state or political subdivision of a state and which agrees to separately account for amounts transferred into such plan from this Plan, or, effective January 1, 2008, a Roth IRA described in Code Section 408A(b), that accepts the Distributee's Eligible Rollover Distribution. However, in the case of an Eligible Rollover Distribution to the surviving spouse or a Participant's surviving Beneficiary, an Eligible Retirement Plan is an individual retirement account or individual retirement annuity. The definition of Eligible Retirement Plan shall also apply in the case of a distribution to a surviving spouse, or to a spouse or former spouse who is the alternate payee under a qualified domestic relation order, as defined in Section 414(p) of the Code. If any portion of an Eligible Rollover Distribution is attributable to payments or distributions from a designated Roth account, an Eligible Retirement Plan with respect to such portion shall include only another designated Roth account of the individual from whose account the payments or distributions were made, or a Roth IRA of such individual. In the case of a nonspouse beneficiary, the direct rollover may be made only to an individual retirement account or annuity described in Code Section 408(a) or 408(b) ("IRA") that is established on behalf of the designated Beneficiary and that will be treated as an inherited IRA pursuant to the provisions of Code Section 402(c)(ii). Further, the determination of any required minimum distribution under Code Section 401(a)(9) that is ineligible for rollover shall be made in accordance with IRS Notice 2007-7, Q&A 17 and 18, 2007-5 I.R.B. 395.
- (iii) "Distributee": A "Distributee" includes a Participant or former Participant. In addition, the Participant's spouse or former Participant's surviving spouse or surviving Beneficiary (effective January 1, 2007) and the Participant's or former Participant's spouse or former spouse who is the alternate payee under a qualified domestic relations order, as defined in Section 414(p) of the Code, are distributees with regard to the interest of the spouse or former spouse.
- (iv) "Direct Rollover": A "Direct Rollover" is a payment by the Plan directly to the Eligible Retirement Plan specified by the Distributee.

7.7 Minimum Distribution Requirements:

(a) General Rules:

- (i) Effective Date. The provisions of this Section will apply for purposes of determining required minimum distributions for calendar years beginning with the 2003 calendar year.
- (ii) Precedence. The requirements of this Section will take precedence over any inconsistent provisions of the Plan.

- (iii) Requirements of Regulations Incorporated. All distributions required under this Section will be determined in accordance with Section 401(a)(9) of the Internal Revenue Code, including the incidental death benefit requirement of Section 401(a)(9)(G), and the Income Tax Regulations thereunder.
- (iv) TEFRA Section 242(b)(2) Elections. Notwithstanding the other provisions of this Section, other than Subsection (iii), distributions may be made under a designation made before January 1, 1984, in accordance with Section 242(b)(2) of the Tax Equity and Fiscal Responsibility Act (TEFRA) and the provisions of the Plan that relate to Section 242(b)(2) of TEFRA.

(b) Time and Manner of Distribution:

- (i) Required Beginning Date. The Participant's entire interest will be distributed, or begin to be distributed, to the Participant no later than the Participant's required beginning date.
- (ii) Death of Participant Before Distributions Begin. If the Participant dies before distributions begin, the Participant's entire interest will be distributed, or begin to be distributed, no later than as follows:
 - (1) If the Participant's surviving spouse is the Participant's sole designated beneficiary, then, except as provided in the adoption agreement, distributions to the surviving spouse will begin by December 31 of the calendar year immediately following the calendar year in which the Participant died, or by December 31 of the calendar year in which the Participant would have attained age 70½, if later.
 - (2) If the Participant's surviving spouse is not the Participant's sole designated beneficiary, then, distributions to the designated beneficiary will begin by December 31 of the calendar year immediately following the calendar year in which the Participant died.
 - (3) If there is no designated beneficiary as of September 30 of the year following the year of the Participant's death, the Participant's entire interest will be distributed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
 - (4) If the Participant's surviving spouse is the Participant's sole designated beneficiary and the surviving spouse dies after the Participant but before distributions to the surviving spouse begin, this Subsection (b)(ii), other than Subsection (b)(ii)(1), will apply as if the surviving spouse were the Participant.

For purposes of this Subsection (ii) and Subsection (e), distributions are considered to begin on the Participant's required beginning date (or, if Subsection (b)(ii)(4) applies, the date distributions are required to begin to the surviving spouse under Subsection (b)(ii)(1)). If annuity payments irrevocably commence to the Participant before the

Participant's required beginning date (or to the Participant's surviving spouse before the date distributions are required to begin to the surviving spouse under Subsection (b)(ii)(1), the date distributions are considered to begin is the date distributions actually commence.

(iii) Form of Distribution. Unless the Participant's interest is distributed in the form of an annuity purchased from an insurance company or in a single sum on or before the required beginning date, as of the first distribution calendar year distributions will be made in accordance with Subsections (c), (d) and (e) of this Section. If the Participant's interest is distributed in the form of an annuity purchased from an insurance company, distributions thereunder will be made in accordance with the requirements of Section 401(a)(9) of the Code and the Treasury regulations.

(c) Determination of Amount to be Distributed Each Year:

- (i) General Annuity Requirements. If the Participant's interest is paid in the form of annuity distributions under the Plan, payments under the annuity will satisfy the following requirements:
 - (1) the annuity distributions will be paid in periodic payments made at intervals not longer than one year;
 - (2) the distribution period will be over a life (or lives) or over a period certain not longer than the period described in Section (d) or (e);
 - (3) once payments have begun over a period certain, the period certain will not be changed even if the period certain is shorter than the maximum permitted;
 - (4) payments will either be nonincreasing or increase only as follows:
 - by an annual percentage increase that does not exceed the annual percentage increase in a cost-of-living index that is based on prices of all items and issued by the Bureau of Labor Statistics;
 - b to the extent of the reduction in the amount of the Participant's payments to provide for a survivor benefit upon death, but only if the beneficiary whose life was being used to determine the distribution period described in Section (d) dies or is no longer the Participant's beneficiary pursuant to a qualified domestic relations order within the meaning of Section 414(p);
 - c to provide cash refunds of employee contributions upon the Participant's death; or
 - d to pay increased benefits that result from a plan amendment.

- (ii) Amount_Required to be Distributed by Required Beginning Date. The amount that must be distributed on or before the Participant's required beginning date (or, if the Participant dies before distributions begin, the date distributions are required to begin under Subsection (b)(ii)(1) or (2)) is the payment that is required for one payment interval. The second payment need not be made until the end of the next payment interval even if that payment interval ends in the next calendar year. Payment intervals are the periods for which payments are received, e.g., bi-monthly, monthly, semi-annually, or annually. All of the Participant's benefit accruals as of the last day of the first distribution calendar year will be included in the calculation of the amount of the annuity payments for payment intervals ending on or after the Participant's required beginning date.
- (iii) Additional Accruals After First Distribution Calendar Year. Any additional benefits accruing to the Participant in a calendar year after the first distribution calendar year will be distributed beginning with the first payment interval ending in the calendar year immediately following the calendar year in which such amount accrues.
- (d) Requirements For Annuity Distributions That Commence During Participant's Lifetime:
- (i) Joint Life Annuities Where the Beneficiary Is Not the Participant's Spouse. If the Participant's interest is being distributed in the form of a joint and survivor annuity for the joint lives of the Participant and a nonspouse beneficiary, annuity payments to be made on or after the Participant's required beginning date to the designated beneficiary after the Participant's death must not at any time exceed the applicable percentage of the annuity payment for such period that would have been payable to the Participant using the table set forth in Q&A-2 of Section 1.401(a)(9)-6T of the Treasury regulations. If the form of distribution combines a joint and survivor annuity for the joint lives of the Participant and a nonspouse beneficiary and a period certain annuity, the requirement in the preceding sentence will apply to annuity payments to be made to the designated beneficiary after the expiration of the period certain.
- Period Certain Annuities. Unless the Participant's spouse is the sole designated beneficiary and the form of distribution is a period certain and no life annuity, the period certain for an annuity distribution commencing during the Participant's lifetime may not exceed the applicable distribution period for the Participant under the Uniform Lifetime Table set forth in Section 1.401(a)(9)-9 of the Treasury regulations for the calendar year that contains the annuity starting date. If the annuity starting date precedes the year in which the Participant reaches age 70, the applicable distribution period for the Participant is the distribution period for age 70 under the Uniform Lifetime Table set forth in Section 1.401(a)(9)-9 of the Treasury regulations plus the excess of 70 over the age of the Participant as of the Participant's birthday in the year that contains the annuity starting date. If the Participant's spouse is the Participant's sole designated beneficiary and the form of distribution is a period certain and no life annuity, the period certain may not exceed the longer of the Participant's applicable distribution period, as determined under this Section (d)(ii), or the joint life and last survivor expectancy of the Participant and the Participant's spouse as determined under the Joint and Last Survivor Table set forth in Section 1.401(a)(9)-9 of the Treasury regulations, using the Participant's and spouse's attained ages as of the Participant's and spouse's birthdays in the calendar year that contains the annuity starting date.

(e) Requirements For Minimum Distributions After the Participant's Death:

(i) Death After Distributions Begin. If the Participant dies after distribution of his or her interest begins in the form of an annuity meeting the requirements of this Article, the remaining portion of the Participant's interest will continue to be distributed over the remaining period over which distributions commenced.

(ii) Death Before Distributions Begin.

- a. Participant Survived by Designated Beneficiary. If the Participant dies before the date distribution of his or her interest begins and there is a designated beneficiary, the Participant's entire interest will be distributed, beginning no later than the time described in Subsection (b)(ii)(1) or (2), over the life of the designated beneficiary or over a period certain not exceeding:
 - (1) unless the annuity starting date is before the first distribution calendar year, the life expectancy of the designated beneficiary determined using the beneficiary's age as of the beneficiary's birthday in the calendar year immediately following the calendar year of the Participant's death; or
 - (2) if the annuity starting date is before the first distribution calendar year, the life expectancy of the designated beneficiary determined using the beneficiary's age as of the beneficiary's birthday in the calendar year that contains the annuity starting date.
- b. No Designated Beneficiary. If the Participant dies before the date distributions begin and there is no designated beneficiary as of September 30 of the year following the year of the Participant's death, distribution of the Participant's entire interest will be completed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
- c. Death of Surviving Spouse Before Distributions to Surviving Spouse Begin. If the Participant dies before the date distribution of his or her interest begins, the Participant's surviving spouse is the Participant's sole designated beneficiary, and the surviving spouse dies before distributions to the surviving spouse begin, this Section (e) will apply as if the surviving spouse were the Participant, except that the time by which distributions must begin will be determined without regard to Subsection (b)(ii)(1).

(f) Definitions:

- (i) Designated Beneficiary. The individual who is designated as the beneficiary under Section 6.5 of the Plan and is the designated beneficiary under Section 401(a)(9) of the Internal Revenue Code and Section 1.401(a)(9)-4 of the Treasury regulations.
- (ii) Distribution Calendar Year. A calendar year for which a minimum distribution is required. For distributions beginning before the Participant's death, the first

distribution calendar year is the calendar year immediately preceding the calendar year which contains the Participant's required beginning date. For distributions beginning after the Participant's death, the first distribution calendar year is the calendar year in which distributions are required to begin pursuant to Subsection (b)(ii).

- (iii) Life Expectancy. Life expectancy as computed by use of the Single Life Table in Section 1.401(a)(9)-9 of the Treasury regulations.
- (iv) Required Beginning Date. The April 1 of the calendar year following the later of the calendar year in which the Participant attains age 70½, or if later, retires.

ARTICLE VIII.

Employment Transfers

8.1 Transfers From This Plan:

- (a) To Another Category With This Employer: If a Participant is employed by the Employer under this Plan and is transferred to employment with this Employer but under another department, classification or category, so that he is no longer eligible to participate in this Plan, such participation shall thereupon cease; his benefits shall remain in the Fund (unless such Participant is eligible for his Normal Pension in accordance with Section 4.1); but he will not continue to accrue service for the purposes of benefit accruals or additional vesting credit for benefits under this Plan. However, if a Participant participates in any other retirement plan sponsored by the Employer within the Fund, he shall continue to accrue Service for vesting purposes under this Plan.
- (b) To Another Municipality: If a Participant's employment by the Employer under this Plan is terminated by virtue of his transfer to employment with another Municipality, his participation in this Plan shall thereupon cease and he shall be subject to the following rules and requirements relating to this Plan and his right and benefits hereunder:
- (1) If he is eligible for a Pension under this Plan as of the date of such employment transfer, he shall be entitled to his Pension. If he is, immediately upon such transfer of employment, covered by the retirement plan under which such other Municipality participates in the Oklahoma Municipal Retirement Fund, and he has not withdrawn his Contribution Accumulation in this Plan, then he will continue to accrue Service in this Plan for the purpose of the determination of attainment of Normal Retirement Age under this Plan, but shall not be entitled to credit for Service while not a Participant under this Plan for the purpose of computing the amount of any benefit under this Plan; or
- (2) If he is not eligible for a Pension under this Plan as of the date of such employment transfer, and he is, immediately upon such transfer of employment, covered by the retirement plan under which such other Municipality participates in the Oklahoma Municipal Retirement Fund, his Contribution Accumulation shall remain in the Fund and will continue to accrue interest, and he will continue to accrue Service for the purpose of meeting eligibility requirements for benefits and for determination of attainment of Normal Retirement Age under this Plan, but shall not be entitled to credit for Service while not a Participant under this Plan for the purpose of computing the amount of any benefit under this Plan and upon so meeting such eligibility requirements for benefits, he or his Beneficiaries shall be entitled to such benefits.

8.2 Transfers to This Plan:

(a) From Another Category with This Employer: Effective for Plan Years beginning on or after July 1, 1998, if a person becomes an Employee and a Participant under this Plan immediately upon his transfer from full-time, regular employment with this Employer under another department, classification or category where he is ineligible for membership only because of the type of such employment, his Service accrued by virtue of such prior employment shall not be counted in determining his eligibility for benefits hereunder and not in computing

the amount of such benefits, and he shall also be subject to all the other provisions of this Plan, provided such transfer occurred prior to the adoption of this Plan. Provided, however, for Plan Years ending prior to July 1, 1998, the rules of the prior Plan document shall apply with respect to such transfers.

- Participant under this Plan immediately upon his transfer from full-time, regular employment with a Municipality other than this Employer, his Service accrued by virtue of such prior employment shall be counted in determining his eligibility and vesting for benefits hereunder, but not in computing the amount of such benefits, and he shall also be subject to all the other provisions of this Plan. An Employee's eligibility for membership under this Plan will be determined by applying the eligibility requirements in the Joinder Agreement as though the date his credited service from the other Municipality began was his date of employment with this Employer. Provided, however, no such Service shall be counted if the Participant was not 100% vested in the other Municipality's qualified retirement plan and the Participant received a distribution of his benefit under such Plan unless the distribution of his benefit was paid after becoming vested with this Employer.
- (c) Previously Fully Vested With Another Municipality: With respect to a Participant who was previously 100% vested in any other Municipality's qualified retirement plan prior to becoming a Participant in this Plan, such Participant's "Service" for purposes of determining years of service for eligibility and vesting under this Plan shall include the Participant's last continuous period during which the Participant was an employee of the other Municipality.
- 8.3 Notice of Transfers: Immediately after any transfer of employment referred to in Sections 8.1 or 8.2, the transferred Employee shall give written notice of such transfer to the Authorized Agent on a form furnished by the Authorized Agent. Such Employee shall not be penalized, however, for failure to give such notice. The Authorized Agent shall give immediate notice in writing of such transfers to the Committee.

<u>ARTICLE IX.</u>

Administration

- 9.1 Administration: The Plan shall be administered by the Committee which is hereby created and established and which shall be composed of the members of the City Council of the Employer. The duties of the Committee shall be performed without compensation other than the compensation, if any, which they receive as officers of the Employer unless additional compensation is specifically provided for by action of the City Council. Any usual and reasonable expenses incurred by the Committee in the administration of this Fund and Plan shall be paid by the Employer.
- (a) Committee: The Committee shall have such powers as may be necessary to discharge its duties hereunder and under the document creating the Oklahoma Municipal Retirement Fund, and under the contract for the pooling of the Fund with similar funds of other Municipalities. Such powers shall include but not be limited to the following powers and duties:
- (1) to delegate to, specify, direct, and supervise the performance of duties of the Authorized Agent, as the agent of the Employer and Committee in matters relating to the Plan, and the Fund, including but not limited to, the duties set forth below in Subsection 9.1(b) and including any duties of the Employer under the Plan, or as set forth in this Subsection 9.1(a);
- (2) acting by direction to the Authorized Agent to file a petition for nomination, or otherwise nominate, and cast the ballot for the election of Trustees of the Oklahoma Municipal Retirement Funds;
- (3) to construe and interpret the Plan and resolve any ambiguities with respect to any of the terms and provisions thereof as written and as applied to the operation of the Plan;
- (4) decide all questions of eligibility and determine the amount, manner and time of payment of any benefits hereunder;
- (5) to prescribe procedures to be followed by Participants in filing applications for benefits:
- (6) to make a determination as to the right of any person to a benefit and to afford any person dissatisfied with such determination the right to a hearing thereon;
- (7) to receive from the Employer, the Trustees, the Trust Service Provider and the Authorized Agent, such information as shall be necessary for the proper administration of the Plan;
- (8) to prepare and distribute, in such manner as it determines to be appropriate, information explaining the Plan;
- (9) to furnish the Employer, upon request, such annual reports with respect to the administration of the Plan as are reasonable and appropriate;

- (10) to receive and review the valuation-report and certification of the Plan, prepared annually by the actuarial firm, and on the basis thereof to certify to the Employer's budgetary authority an appropriate contribution rate in time for the incorporation, when necessary, of the resulting costs in the budget, and make timely appropriations therefor;
- (11) to receive and review reports from the auditor appointed by the Trustees, the City Treasurer and City Auditors, of the financial condition of the Fund;
- (12) to have full power, to manage and control, the Plan and Fund and to authorize in writing, all payments from the Fund by written direction of the Authorized Agent, or otherwise; and
- (13) to sue in any court of competent jurisdiction for the enforcement of any contract, claim or other right, and to defend against or to compromise, settle or otherwise dispose of any claim or suit against the Employer, the Plan, or the City Treasurer, as Treasurer of the Plan.
 - (14) to appoint such person or persons as necessary to perform the following:
 - a. to receive and separately account for, payments, appropriations, apportionments, allocations, payroll deductions, and any other assets, which are for, or consist of contributions or assets under the Plan for the Fund, which are made by the Employer, the Participants, or from any other source;
 - b. to transfer, remit, pay over and deliver, upon the written direction of the Authorized Agent, as soon as practicable after his receipt thereof, all such contributions and assets, to the Oklahoma Municipal Retirement Fund for management and investment:
 - c. to keep as evidence and permanent records, all such written directions of the Authorized Agent for such transfers and disbursements, maintain accurate accounts and records of such receipts, transfers and disbursements, and keep such other records and furnish such information and advice to the Employer, the City Council, the Committee and the Authorized Agent as may be necessary and proper for the performance of such duties in coordinating the administration and operation of the Plan;
 - d. maintain such records including vital statistics on health, age, sex, birth, death, Compensation and length of Service of all the Participants of the Employer or their beneficiaries who are included in the Plan or who are, or may become eligible for such inclusion, as are necessary for the proper administration of the Plan, and furnish such information as is requested by the Authorized Agent, or is requested by the Administrator;
 - e. notify the Authorized Agent when any Participant is eligible for Retirement under the Plan; and
 - f. attend meetings of the Committee while matters pertaining to the Plan, the Participants or their beneficiaries are under consideration.

The Committee shall have no power to waive or fail to apply any requirements of eligibility for a Pension under the Plan. The Committee may adopt such rules, regulations and actuarial tables as it deems necessary or desirable to administer the Plan. All such rules, regulations and decisions shall be uniformly and consistently applied to all Employees in similar circumstances.

Any such rule or decision which is not inconsistent with the provisions of the Plan shall be conclusive and binding upon all persons affected by it and there shall be no appeal from any ruling by the Committee which is within its authority.

When making a determination or calculation, the Committee shall be entitled to rely upon information furnished by the Trustees, the Trust Service Provider, the Employer, the Authorized Agent, the legal counsel of the Employer, or the actuary for the Plan.

- (b) Authorized Agent: An Authorized Agent shall be designated in writing by the Committee and shall act as the agent of the Employer (but not the agent of the Trustees or the Trust Service Provider of the Fund) in matters pertaining to the Plan and the Fund, to centralize in one person the local administration and coordination thereof, and to file payroll and contribution information, to file claims, forms and applications for Participants, and to advise Participants, the Employer and the Committee. The Authorized Agent, under the control and direction of the Committee, shall have such general duties as the Employer and the Committee may deem necessary and proper for such purposes, which duties shall include but not be limited to, the following:
- (1) to coordinate the deduction of Participant contributions and to see that Employer and Participant contributions are properly received and forwarded promptly to the Fund for management and investment;
- (2) to forward any communications directed to Participants and Beneficiaries by the Trustees, the Trust Service Provider or the Fund;
- (3) to lend assistance to Participants and Beneficiaries in filing applications for benefits, and in communicating with the Employer, the Committee and the Trustees or the Trust Service Provider of the Fund and to forward such communications to the addressees;
- (4) to keep the Employer and Committee informed regarding Employer contribution rates and funds required to meet the costs of the Plan;
- (5) to assist the Committee in determining whether or not Employees are eligible for participation in the Plan;
- (6) to certify at the direction of the Committee that an Employee is on an Authorized Leave of Absence, paid or unpaid; and
- (7) to file at the direction of the Committee a petition or nomination, and cast a ballot for election of Trustees of the Fund.

- (c) Plan Municipal Counselor: The Committee of the Employer shall appoint the legal advisor of the Employer and the Committee, and such legal advisor shall represent them in any legal matters, proceedings, or litigation.
- 9.2 Bonds: No bond to secure the performance of administrative duties in the operation of the Plan and the Fund, shall be required of any persons or organizations unless required by law, or unless required by the Trust indenture establishing Fund, or unless required by the Employer for any persons or organizations engaged in the administration of the Plan. If such a bond is required by law, the Trustees or the Employer, the premiums therefor shall be paid as expenses of the Fund. Any agents, officials of employees of the Employer engaged in the administration for the Plan shall be covered as to the performance of such administrative duties, by any official or other bond covering their regular duties otherwise.
- 9.3 Benefit Payments: All benefits which are to be paid pursuant to the provisions of the Plan, shall be paid under the direction of the Committee out of the applicable portion of the Fund, upon written directions of the Committee acting through the Authorized Agent.

9.4 Abandonment of Benefits:

- (a) If, anytime following the date either of a Participant or Beneficiary of a deceased Participant becomes entitled to receive any non-deferred benefits under the Plan, then, if the whereabouts of such Participant or Beneficiary is unknown, the benefits may be forfeited in certain limited circumstances as provided hereafter. If the Committee has mailed to the Participant or Beneficiary notice of the present right to receive benefits, and the Committee mails such notice again after one year, then, if no claim has been received by the second anniversary of the first mailing of the notice, the Accounts representing unclaimed Benefits (including those holding Participant contributions) shall be forfeited.
- (b) Each Participant and Beneficiary shall file with the Committee, from time to time in writing, their post office address and each change of post office address, if any, and the Committee shall not be obliged to search for or ascertain the whereabouts of any Participant or Beneficiary. Any communication addressed to a Participant or Beneficiary at their last post office address filed with the Committee, or if no such address was filed, then at their last post office address as shown on the Employer's records, shall be binding on the Participant and the Beneficiary for all purposes of the Plan and Trust.
- (c) In the event that the whereabouts of a lost Participant, or lost Beneficiary of a deceased Participant, ever becomes known to the Committee, and either of such parties makes a claim for benefits, the Committee shall, if the Plan is in existence, reinstate any Benefits which have been previously forfeited to satisfy such claim, including any applicable cost-of-living adjustments. For purposes of this Subsection (c), the limitations under Section 415 of the Code shall not apply.
- 9.5 Benefits Payable to Incompetents: Any payments due hereunder to a minor or other person under legal disability may be made; at the discretion of the Committee, (i) to a parent, spouse, relative by blood or marriage, or (ii) the legal representative of the said person. The Committee shall not be required to see to the application of any such payment, and the payee's

receipt shall be a full and final discharge of all responsibility hereunder of the Employer, the Committee and the Trustees.

9.6 No Participant Loans Permitted: Loans to Participants are not permitted under this Plan. Any references to Participant Loans in the Trust Indenture establishing the Oklahoma Municipal Retirement Fund shall not be applicable to this Plan.

ARTICLE X.

Limitations

10.1 Limitations on Benefits Relating to Section 415 of Internal Revenue Code of 1986:

- (a) Notwithstanding any other provision contained herein to the contrary, the benefits payable to a Participant from the Plan provided by employer contributions (including contributions picked up by the employer under Code Section 414(h)) shall be subject to the limitations of Code Section 415 in accordance with the provisions of this Article. The limitations of this Article shall apply in limitation years beginning on or after July 1, 2007, except as otherwise provided below.
- (b) Except as provided below, effective for limitation years ending after December 31, 2007, any accrued retirement benefit otherwise payable to a Participant under the Plan at any time shall not exceed the Maximum Permissible Benefit. If the benefit the Participant would otherwise accrue in a Limitation Year would produce an Annual Benefit in excess of the Maximum Permissible Benefit, the benefit shall be limited (or the rate of accrual reduced) to a benefit that does not exceed the Maximum Permissible Benefit.
- (c) If the Participant is, or has ever been, a Participant in another qualified defined benefit plan (without regard to whether the Plan has been terminated) maintained by the Employer or a predecessor employer, the sum of the Participant's Annual Benefits from all such plans may not exceed the Maximum Permissible Benefit. Where the Participant's Employer-provided benefits under all such defined benefit plans (determined as of the same age) would exceed the Maximum Permissible Benefit applicable at that age, the Employer will apply the benefit accrual limit first to the plan that is not a broad-based participation plan.
- (d) The application of the provisions of this Article shall not cause the Maximum Permissible Benefit for any Participant to be less than the Participant's accrued benefit under all the defined benefit plans of the Employer or a predecessor employer as of the end of the last Limitation Year beginning before July 1, 2007 under provisions of the plans that were both adopted and in effect before April 5, 2007. The preceding sentence applies only if the provisions of such defined benefit plans that were both adopted and in effect before April 5, 2007 satisfied the applicable requirements of statutory provisions, regulations, and other published guidance relating to Code Section 415 in effect as of the end of the last Limitation Year beginning before July 1, 2007, as described in Treas. Reg. Section 1.415(a)-1(g)(4).
- (e) The limitations of this Article shall be determined and applied taking into account the rules in Section 10.7.
- 10.2 **Definitions:** For purposes of calculating the limitations which are otherwise applicable to a Participant pursuant to Section 415 of the Code and this Plan, the following definitions shall apply.
- (a) Annual Benefit: A benefit that is payable annually in the form of a straight life annuity. Except as provided below, where a benefit is payable in a form other than a straight life annuity, the benefit shall be adjusted to an actuarially equivalent straight life annuity that begins

at the same time as such other form of benefit and is payable on the first day of each month before applying the limitations of this Article. For a Participant who has or will have distributions commencing at more than one annuity starting date, the Annual Benefit shall be determined as of each such annuity starting date (and shall satisfy the limitations of this Article as of each such date), actuarially adjusting for past and future distributions of benefits commencing at the other annuity starting dates. For this purpose, the determination of whether a new starting date has occurred shall be made without regard to Treas. Reg. Section 1.401(a)-20. Q&A 10(d), and with regard to Treas. Reg. Section 1.415(b)-1(b)(1)(iii)(B) and (C).

No actuarial adjustment to the benefit shall be made for (a) survivor benefits payable to a surviving spouse under a qualified joint and survivor annuity to the extent such benefits would not be payable if the Participant's benefit were paid in another form; (b) benefits that are not directly related to retirement benefits (such as a qualified disability benefit, preretirement incidental death benefits, and postretirement medical benefits); or (c) the inclusion in the form of benefit of an automatic benefit increase feature, provided the form of benefit is not subject to Code Section 417(e)(3) and would otherwise satisfy the limitations of this Article, and the Plan provides that the amount payable under the form of benefit in any Limitation Year shall not exceed the limits of this Article applicable at the annuity starting date, as increased in subsequent years pursuant to Code Section 415(d). For this purpose, an automatic benefit increase feature is included in a form of benefit if the form of benefit provides for automatic, periodic increases to the benefits paid in that form.

The determination of the Annual Benefit shall take into account Social Security supplements described in Code Section 411(a)(9) and benefits transferred from another defined benefit plan, other than transfers of distributable benefits pursuant to Treas. Reg. Section 1.411(d)-4, Q&A-3(c), but shall disregard benefits attributable to employee contributions or rollover contributions. Effective for distributions in Plan Years beginning after December 31, 2003, the determination of actuarial equivalence of forms of benefit other than a straight life annuity shall be made in accordance with Section 10.2(a)(1) or Section 10.2(a)(2).

- (1) Benefit Forms Not Subject to §417(e)(3): The straight life annuity that is actuarially equivalent to the Participant's form of benefit shall be determined under this Section 10.2(a)(1) if the form of the Participant's benefit is either (1) a nondecreasing annuity (other than a straight life annuity) payable for a period of not less than the life of the Participant (or, in the case of a qualified pre-retirement survivor annuity, the life of the surviving spouse), or (2) an annuity that decreases during the life of the Participant merely because of (a) the death of the survivor annuitant (but only if the reduction is not below 50% of the benefit payable before the death of the survivor annuitant), or (b) the cessation or reduction of Social Security supplements or qualified disability payments (as defined in Code Section 401(a)(11)).
- (i) <u>Limitation Years beginning before July 1, 2007</u>. For Limitation Years beginning before July 1, 2007, the actuarially equivalent straight life annuity is equal to the annual amount of the straight life annuity commencing at the same annuity starting date that has the same actuarial present value as the Participant's form of benefit computed using whichever of the following produces the greater annual amount: (I) the interest rate specified in Section 2.1(b) of the Plan and the mortality table (or other tabular factor) specified in Section 2.1(b) of

the Plan for adjusting benefits in the same form; and (II) a 5 percent interest rate assumption and the applicable mortality table defined in Section 2.1(b) of the Plan for that annuity starting date.

- (ii) <u>Limitation Years beginning on or after July 1, 2007</u>. For Limitation Years beginning on or after July 1, 2007, the actuarially equivalent straight life annuity is equal to the greater of (1) the annual amount of the straight life annuity (if any) payable to the Participant under the Plan commencing at the same annuity starting date as the Participant's form of benefit; and (2) the annual amount of the straight life annuity commencing at the same annuity starting date that has the same actuarial present value as the Participant's form of benefit, computed using a 5 percent interest rate assumption and the applicable mortality table defined in Section 2.1(b) of the Plan for that annuity starting date.
- (2) Benefit Forms Subject to Code Section 417(e)(3): The straight life annuity that is actuarially equivalent to the Participant's form of benefit shall be determined under this paragraph if the form of the Participant's benefit is other than a benefit form described in Section 10.2(a)(1). In this case, the actuarially equivalent straight life annuity shall be determined as follows:
- (i) Annuity Starting Date in Plan Years Beginning After 2005. Except as provided in Section 10.2(a)(2)(iii), if the annuity starting date of the Participant's form of benefit is in a Plan Year beginning after December 31, 2005, the actuarially equivalent straight life annuity is equal to the greatest of:
 - (a) The annual amount of the straight life annuity commencing at the same annuity starting date that has the same actuarial present value as the Participant's form of benefit, computed using the interest rate and the mortality table (or other tabular factor) specified in the Plan for adjusting benefits in the same form;
 - (b) The annual amount of the straight life annuity commencing at the same annuity starting date that has the same actuarial present value as the Participant's form of benefit, computed using a 5.5 percent interest rate assumption and the applicable mortality table for the distribution under Regulations Section 1.417(e)-1(d)(2) (determined in accordance with Section 10.2(a)(2)(v) for Plan Years after the effective date specified below); and
 - (c) The annual amount of the straight life annuity commencing at the same annuity starting date that has the same actuarial present value as the Participant's form of benefit, computed using the applicable interest rate for the distribution under Regulations Section 1.417(e)-1(d)(3) (determined in accordance with Section 10.2(a)(2)(v) for Plan Years on or after January 1, 2008 and the applicable mortality table for the distribution under Regulations Section 1.417(e)-1(d)(2) (determined in accordance with Section 10.2(a)(2)(v) for Plan Years after the effective date specified below), divided by 1.05.

The effective date of the applicable mortality table above is for years beginning after December 31, 2008.

(ii) Annuity Starting Date in Plan Years Beginning in 2004 or 2005. If the annuity starting date of the Participant's form of benefit is in a Plan Year beginning in 2004 or 2005, the actuarially equivalent straight life annuity is equal to the annual amount of the straight life annuity commencing at the same annuity starting date that has the same actuarial present value as the Participant's form of benefit, computed using whichever of the following produces the greater annual amount: (I) the interest rate specified in Section 2.1(b) of the Plan and the mortality table (or other tabular factor) specified in Section 2.1(b) of the Plan for adjusting benefits in the same form; and (II) a 5.5 percent interest rate assumption and the applicable mortality table for the distribution under Treas. Reg. Section 1.417(e)-1(d)(2).

However, this Section does not supersede any prior election to apply the transition rule of section 101(d)(3) of PFEA as described in Notice 2004-78.

- (iii) Annuity Starting Date in small plans for Plan Years Beginning in 2009 and later. If the annuity starting date of the Participant's form of benefit is in a Plan Year beginning in or after 2009, and if the Plan is maintained by an eligible employer as defined Code Section 408(p)(2)(C)(i), the actuarially equivalent straight life annuity is equal to the annual amount of the straight life annuity commencing at the same annuity starting date that has the same actuarial present value as the Participant's form of benefit, computed using whichever of the following produces the greater annual amount:
 - (a) The interest rate and the mortality table (or other tabular factor) specified in the Plan for adjusting benefits in the same form; and
 - (b) A 5.5 percent interest rate assumption and the applicable mortality table described in Section 10.2(a)(2)(iv).
- (iv) Definition of "Applicable Mortality Table." Effective for annuity starting dates in a Plan Year beginning on or after January 1, 2008, for purposes of this Article, the "applicable mortality table" means the applicable mortality table within the meaning of Code Section 417(e)(3)(B) as described in Revenue Ruling 2007-67.
- (v) Applicable interest rate. For purposes of the Plan's provisions relating to the calculation of the present value of a benefit payment that is subject to Code Section 417(e), as well as any other Plan provision referring directly or indirectly to the "applicable interest rate" or "applicable mortality table" used for purposes of Code Section 417(e), any provision prescribing the use of the annual rate of interest on 30-year U.S. Treasury securities shall be implemented by instead using the rate of interest determined by applicable interest rate described by Code Section 417(e) after its amendment by PPA. Specifically, the applicable interest rate shall be the adjusted first, second, and third segment rates applied under the rules similar to the rules of Code Section 430(h)(2)(C) for the calendar month (lookback month) before the first day of the Plan Year in which the annuity starting date occurs (stability period). For this purpose, the first, second, and third segment rates are the first, second, and third segment rates which would be determined under Code Section 430(h)(2)(C) if:

- (a) Code Section 430(h)(2)(D) were applied by substituting the average yields for the month described in the preceding paragraph for the average yields for the 24-month period described in such section, and
- (b) Code Section 430(h)(2)(G)(i)(II) were applied by substituting "Section 417(e)(3)(A)(ii)(II) for "Section 412(b)(5)(B)(ii)(II)," and
- (c) The applicable percentage under Code Section 430(h)(2)(G) is treated as being 20% in 2008, 40% in 2009, 60% in 2010, and 80% in 2011.
- (b) Compensation: Compensation is defined as wages, salaries, and fees for professional services and other amounts received (without regard to whether or not an amount is paid in cash) for personal services actually rendered in the course of employment with the employer maintaining the Plan to the extent that the amounts are includible in gross income (including, but not limited to, commissions paid salespersons, compensation for services on the basis of a percentage of profits, commissions on insurance premiums, tips, bonuses, fringe benefits, and reimbursements, or other expense allowances under a nonaccountable plan (as described in Treas. Reg. Section § 1.62-2(c), and excluding the following:
- (i) Employer contributions (other than elective contributions described in Code Sections 402(e)(3), 408(k)(6), 408(p)(2)(A)(i), or 457(b)) to a plan of deferred compensation (including a simplified employee pension described in Code Section 408(k) or a simple retirement account described in Code Section 408(p), and whether or not qualified) to the extent such contributions are not includible in the employee's gross income for the taxable year in which contributed, and any distributions (whether or not includible in gross income when distributed) from a plan of deferred compensation (whether or not qualified);
- (ii) Amounts realized from the exercise of a nonstatutory stock option (that is, an option other than a statutory stock option as defined in Treas. Reg. Section 1.421-1(b)), or when restricted stock (or property) held by the employee either becomes freely transferable or is no longer subject to a substantial risk of forfeiture:
- (iii) Amounts realized from the sale, exchange or other disposition of stock acquired under a statutory stock option;
- (iv) Other amounts that receive special tax benefits, such as premiums for group-term life insurance (but only to the extent that the premiums are not includible in the gross income of the employee and are not salary reduction amounts that are described in Code Section 125);
- (v) Other items of remuneration that are similar to any of the items listed in (i) through (iv).

For Limitation Years beginning more than 90 days after the close of the first regular legislative session of the legislative body with authority to amend the Plan that begins on or after July 1, 2007, compensation for a Limitation Year shall also include compensation paid by the later of 2½ months after an employee's severance from employment with the employer maintaining the Plan or the end of the Limitation Year that includes the date of the employee's severance from

employment with the employer maintaining the Plan, if the payment is regular compensation for services during the employee's regular working hours, or compensation for services outside the employee's regular working hours (such as overtime or shift differential), commissions, bonuses, or other similar payments, and, absent a severance from employment, the payments would have been paid to the employee while the employee continued in employment with the employer.

Any payments not described above shall not be considered compensation if paid after severance from employment, even if they are paid by the later of 2½ months after the date of severance from employment or the end of the Limitation Year that includes the date of severance from employment, except, payments to an individual who does not currently perform services for the employer by reason of qualified military service (within the meaning of Code Section 414(u)(1)) to the extent these payments do not exceed the amounts the individual would have received if the individual had continued to perform services for the employer rather than entering qualified military service.

Back pay, within the meaning of Treas. Reg. Section 1.415(c)-2(g)(8), shall be treated as compensation for the Limitation Year to which the back pay relates to the extent the back pay represents wages and compensation that would otherwise be included under this definition.

For Limitation Years beginning after December 31, 1997, or Compensation paid or made available during such Limitation Year shall include amounts that would otherwise be included in compensation but for an election under Code Sections 125(a), 402(e)(3), 402(h)(1)(B), 402(k), or 457(b).

For Limitation Years beginning after December 31, 2000, Compensation shall also include any elective amounts that are not includible in the gross income of the employee by reason of Code Section 132(f)(4).

For Limitation Years beginning after December 31, 2001, Compensation shall also include deemed Code Section 125 Compensation. Deemed Code Section 125 compensation is an amount that is excludable under Code Section 106 that is not available to a Participant in cash in lieu of group health coverage under a Code Section 125 arrangement solely because the Participant is unable to certify that he or she has other health coverage. Amounts are deemed Code Section 125 compensation only if the employer does not request or otherwise collect information regarding the Participant's other health coverage as part of the enrollment process for the health plan.

- (c) Defined Benefit Dollar Limitation: Effective for Limitation Years ending after December 31, 2001, the Defined Benefit Dollar Limitation is \$160,000, automatically adjusted under Code Section 415(d) for increases in the cost-of-living, effective January 1 of each year, as published in the Internal Revenue Bulletin, and payable in the form of a straight life annuity. The new limitation shall apply to Limitation Years ending with or within the calendar year of the date of the adjustment, but a Participant's benefits shall not reflect the adjusted limit prior to January I of that calendar year.
- (d) Employer: For purposes of this Article, employer shall mean the employer that adopts this plan, and all members of a controlled group of corporations, as defined in Code Section 414(b), as modified by Code Section 415(h)), all commonly controlled trades or

businesses (as defined in Code Section 414(c), as modified, except in the case of a brother-sister group of trades or businesses under common control, by Code Section 415(h)), or affiliated service groups (as defined in Code Section 414(m)) of which the adopting employer is a part, and any other entity required to be aggregated with the employer pursuant to Code Section 414(o).

- (e) Formerly Affiliated Plan of the Employer: A plan that, immediately prior to the cessation of affiliation, was actually maintained by the employer and, immediately after the cessation of affiliation, is not actually maintained by the employer. For this purpose, cessation of affiliation means the event that causes an entity to no longer be considered the employer, such as the sale of a member controlled group of corporations, as defined in Code Section 414(b), as modified by Code Section 415(h), to an unrelated corporation, or that causes a plan to not actually be maintained by the employer, such as transfer of plan sponsorship outside a controlled group.
- (f) Limitation Year: The Plan Year: All qualified plans maintained by the employer must use the same Limitation Year. If the Limitation Year is amended to a different 12-consecutive month period, the new Limitation Year must begin on a date within the Limitation Year in which the amendment is made.
- (g) Maximum Permissible Benefit: The Defined Benefit Dollar Limitation (adjusted where required, as provided below).
- (1) Adjustment for Less Than 10 Years of Participation or Service: If the Participant has less than 10 years of participation with the Employer, the Defined Benefit Dollar Limitation shall be multiplied by a fraction -- (i) the numerator of which is the number of Years (or part thereof, but not less than one year) of Participation in the Plan, and (ii) the denominator of which is 10.
- (2) Adjustment of Defined-Benefit-Dollar Limitation for Benefit Commencement Before Age 62 or after Age 65: Effective for benefits commencing in Limitation Years ending after December 31, 2001, the Defined Benefit Dollar Limitation shall be adjusted if the annuity starting date of the Participant's benefit is before age 62 or after age 65. If the annuity starting date is before age 62, the Defined Benefit Dollar Limitation shall be adjusted under Subsection (g)(2)(i), as modified by (g)(2)(iii). If the annuity starting date is after age 65, the Defined Benefit Dollar Limitation shall be adjusted under Subsection (g)(2)(ii), as modified by, Subsection (g)(2)(iii),
- (i) Adjustment of Defined Benefit Dollar Limitation for Benefit Commencement Before Age 62.
- I. Limitation Years Beginning Before July 1, 2007. If the annuity starting date for the Participant's benefit is prior to age 62 and occurs in a Limitation Year beginning before July 1, 2007, the Defined Benefit Dollar Limitation for the Participant's annuity starting date is the annual amount of a benefit payable in the form of a straight life annuity commencing at the Participant's annuity starting date that is the actuarial equivalent of the Defined Benefit Dollar Limitation (adjusted under Subsection (g)(1) above for years of participation less than 10, if required) with actuarial equivalence computed using whichever of

the following produces the smaller annual amount: (1) the interest rate specified in Section 2.1(b) of the Plan and the mortality table (or other tabular factor) specified in Section 2.1(b) of the Plan; or (2) a 5-percent interest rate assumption and the applicable mortality table as defined in Section 2.1(b) of the Plan.

II. Limitation Years Beginning on or After July 1, 2007.

- A. Plan Does Not Have Immediately Commencing Straight Life Annuity Payable at Both Age 62 and the Age of Benefit Commencement. If the annuity starting date for the Participant's benefit is prior to age 62 and occurs in a Limitation Year beginning on or after July 1, 2007, and the Plan does not have an immediately commencing straight life annuity payable at both age 62 and the age of benefit commencement, the Defined Benefit Dollar Limitation for the Participant's annuity starting date is the annual amount of a benefit payable in the form of a straight life annuity commencing at the Participant's annuity starting date that is the actuarial equivalent of the Defined Benefit Dollar Limitation (adjusted under Section 10.2(g)(1) for years of participation less than 10, if required) with actuarial equivalence computed using a 5 percent interest rate assumption and the applicable mortality table for the annuity starting date as defined in Section 2.1(b) of the Plan (and expressing the Participant's age based on completed calendar months as of the annuity starting date).
- B. Plan Has Immediately Commencing Straight Life Annuity Payable at Both Age 62 and the Age of Benefit Commencement. If the annuity starting date for the Participant's benefit is prior to age 62 and occurs in a Limitation Year beginning on or after July 1, 2007, and the Plan has an immediately commencing straight life annuity payable at both age 62 and the age of benefit commencement, the Defined Benefit Dollar Limitation for the Participant's annuity starting date is the lesser of the limitation determined under Section 10.2(g)(2)(i)(II)(A), and the Defined Benefit Dollar Limitation (adjusted under Section 10.2(g)(1) for years of participation less than 10, if required) multiplied by the ratio of the annual amount of the immediately commencing straight life annuity under the Plan at the Participant's annuity starting date to the annual amount of the immediately commencing straight life annuity under the Plan at age 62, both determined without applying the limitations of this Article.
- (ii) Adjustment of Defined Benefit Dollar Limitation for Benefit Commencement After Age 65:
- I. Limitation Years Beginning Before July 1, 2007. If the annuity starting date for the Participant's benefit is after age 65 and occurs in a Limitation Year beginning before July 1, 2007, the Defined Benefit Dollar Limitation for the Participant's annuity starting date is the annual amount of a benefit payable in the form of a straight life annuity commencing at the Participant's annuity starting date that is the actuarial equivalent of the Defined Benefit Dollar Limitation (adjusted under subsection (g)(1) above for years of participation less than 10, if required) with actuarial equivalence computed using whichever of the following produces the smaller annual amount: (1) the-interest rate specified in Section 2.1(b) of the Plan and the mortality table (or other tabular factor) specified in Section 2.1(b) of the Plan; or (2) a 5-percent interest rate assumption and the applicable mortality table as defined in Section 2.1(b) of the Plan.

II. Limitation Years Beginning On or After July 1, 2007.

- A. Plan Does Not Have Immediately Commencing Straight Life Annuity Payable at Both Age 65 and the Age of Benefit Commencement. If the annuity starting date for the Participant's benefit is after age 65 and occurs in a Limitation Year beginning on or after July 1, 2007, and the Plan does not have an immediately commencing straight life annuity payable at both age 65 and the age of benefit commencement, the Defined Benefit Dollar Limitation at the Participant's annuity starting date is the annual amount of a benefit payable in the form of a straight life annuity commencing at the Participant's annuity starting date that is the actuarial equivalent of the Defined Benefit Dollar Limitation (adjusted under Subsection (g)(1) for years of participation less than 10, if required), with actuarial equivalence computed using a 5 percent interest rate assumption and the applicable mortality table for that annuity starting date as defined in Section 2.1(b) of the Plan (and expressing the Participant's age based on completed calendar months as of the annuity starting date).
- Plan Has Immediately Commencing Straight Life Annuity Payable at Both Age 65 and the Age of Benefit Commencement. If the annuity starting date for the Participant's benefit is after age 65 and occurs in a Limitation Year beginning on or after July 1, 2007, and the Plan has an immediately commencing straight life annuity payable at both age 65 and the age of benefit commencement, the Defined Benefit Dollar Limitation at the Participant's annuity starting date is the lesser of the limitation determined under Subsection (g)(2)(ii)II.A., and the Defined Benefit Dollar Limitation (adjusted under Subsection (g)(1) for years of participation less than 10, if required) multiplied by the ratio of the annual amount of the adjusted immediately commencing straight life annuity under the Plan at age 65, both determined without applying the limitations of this Article. For this purpose, the adjusted immediately commencing straight life annuity under the Plan at the Participant's annuity starting date is the annual amount of such annuity payable to the Participant, computed disregarding the Participant's accruals after sixty-five (65) years of age but including actuarial adjustments even if those actuarial adjustments are used to offset accruals; and the adjusted immediately commencing straight life annuity under the Plan at sixty-five (65) years of age is the annual amount of such annuity that would be payable under the Plan to a hypothetical member who is sixty-five (65) years of age and has the same accrued benefit as the Participant.
- (iii) Notwithstanding the other requirements of this Subsection (g)(2), in adjusting the Defined Benefit Dollar Limitation for the Participant's annuity starting date under Subsections (g)(2)(i)I. and (g)(2)(i)II.A., (g)(2)(ii)I., (g)(2)(ii)II.A., no adjustment shall be made to the Defined Benefit Dollar Limitation to reflect the probability of a Participant's death between the annuity starting date and age 62, or between age 65 and the annuity starting date, as applicable, if benefits are not forfeited upon the death of the Participant prior to the annuity starting date. To the extent benefits are forfeited upon death before the annuity starting date, such an adjustment shall be made. For this purpose, no forfeiture shall be treated as occurring upon the Participant's death if the Plan does not charge Participants for providing a qualified preretirement survivor annuity, as defined in Code Section 417(c), upon the Participant's death.
- (iv) Notwithstanding any other provision to the contrary, for limitation years beginning on or after January 1, 1997, if payment begins before the Participant reaches sixty-two

- (62) years of age, the reductions in the limitations in this subsection shall not apply to a Participant who is a "qualified participant" as defined in Section 415(b)(2)(H) of the Code.
- (3) Minimum benefit permitted: Notwithstanding anything else in this section to the contrary, the benefit otherwise accrued or payable to a Participant under this plan shall be deemed not to exceed the Maximum Permissible Benefit if:
- (i) the retirement benefits payable for a Limitation Year under any form of benefit with respect to such Participant under this plan and under all other defined benefit plans (without regard to whether a plan has been terminated) ever maintained by the employer do not exceed \$10,000 multiplied by a fraction (I) the numerator of which is the Participant's number of Years (or part thereof, but not less than one year) of Service (not to exceed 10) with the employer, and (II) the denominator of which is 10; and
- (ii) the employer (or a predecessor employer) has not at any time maintained a defined contribution plan in which the Participant participated (for this purpose, mandatory employee contributions under a defined benefit plan, individual medical accounts under Code Section 401(h), and accounts for postretirement medical benefits established under Code Section 419A(d)(1) are not considered a separate defined contribution plan).
- (4) For limitation years beginning on or after January 1, 1995, subsection (g)(1) of this Section, subsection (g)(1)(i) of this Section, and the proration provided under subsection (g)(3)(i) of this Section, shall not apply to a benefit paid under the Plan as a result of the Participant becoming disabled by reason of personal inquiries or sickness or amounts received by the beneficiaries, survivors or estate of the Participant as a result of the death of the Participant.
- (5) Effective for years beginning after December 31, 1997, if a member repays to the Plan any amounts received because of the Participant's prior termination pursuant to Section 10.11 of this Plan, such repayment shall not be taken into account for purposes of Section 415 of the Code pursuant to Code Section 415(k)(3).
- (6) For distributions made in limitation years beginning on or after January 1, 2000, the combined limit of repealed Code Section 415(e) shall not apply.
- 10.3 Predecessor Employer: If the employer maintains a plan that provides a benefit which the Participant accrued while performing services for a former employer, the former employer is a predecessor employer with respect to the Participant in the Plan. A former entity that antedates the employer is also a predecessor employer with respect to a Participant if, under the facts and circumstances, the employer constitutes a continuation of all or a portion of the trade or business of the former entity.
- 10.4 Severance from Employment: An employee has a severance from employment when the employee ceases to be an employee of the employer maintaining the Plan. An employee does not have a severance from employment if, in connection with a change of employment, the employee's new employer maintains the Plan with respect to the employee.
- 10.5 Year of Participation: The Participant shall be credited with a Year of Participation (computed to fractional parts of a year) for each accrual computation period for which the

following conditions are met: (1) the Participant is credited with at least the number of hours of service (or period of service if the elapsed time method is used) for benefit accrual purposes, required under the terms of the Plan in order to accrue a benefit for the accrual computation period, and (2) the Participant is included as a Participant under the eligibility provisions of the Plan for at least one day of the accrual computation period. If these two conditions are met, the portion of a year of participation credited to the Participant shall equal the amount of benefit accrual service credited to the Participant for such accrual computation period. A Participant who is permanently and totally disabled within the moaning of Code Section 415(c)(3)(C)(i) for an accrual computation period shall receive a Year of Participation with respect to that period. In addition, for a Participant to receive a Year of Participation (or part thereof) for an accrual computation period, the Plan must be established no later that the last day of such accrual computation period. In no event shall more than one Year of Participation be credited for any 12-month period.

10.6 Year of Service: For purposes of Section 10.2(g), the Participant shall be credited with a Year of Service (computed to fractional parts of a year) for each accrual computation period for which the Participant is credited with at least the number of hours of service (or period of service if the elapsed time method is used) for benefit accrual purposes, required under the terms of the Plan in order to accrue a benefit for the accrual computation period, taking into account only service with the employer or a predecessor employer.

10.7 Other Rules:

- (a) Benefits Under Terminated Plans: If a defined benefit plan maintained by the employer has terminated with sufficient assets for the payment of benefit liabilities of all plan Participants and a Participant in the Plan has not yet commenced benefits under the Plan, the benefits provided pursuant to the annuities purchased to provide the Participant's benefits under the terminated plan at each possible annuity starting date shall be taken into account in applying the limitations of this Article. If there are not sufficient assets for the payment of all Participants' benefit liabilities, the benefits taken into account shall be the benefits that are actually provided to the Participant under the terminated plan.
- benefits Transferred From the Plan: If a Participant's benefits under a defined benefit plan maintained by the employer are transferred to another defined benefit plan maintained by the employer and the transfer is not a transfer of distributable benefits pursuant Treas. Reg. Section 1.411(d)-4, Q&A-3(c), the transferred benefits are not treated as being provided under the transferor plan (but are taken into account as benefits provided under the transferee plan). If a Participant's benefits under a defined benefit plan maintained lay the employer are transferred to another defined benefit plan that is not maintained by the employer and the transfer is not a transfer of distributable benefits pursuant to Treas. Reg. Section 1.411(d)-4, Q&A-3(c), the transferred benefits are treated by the employer's plan as if such benefits were provided under annuities purchased to provide benefits under a plan maintained by the employer that terminated immediately prior to the transfer with sufficient assets to pay all Participants' benefit liabilities under the Plan. If a Participant's benefits under a defined benefit plan in a transfer of distributable benefits pursuant Treas. Reg. Section 1.411(d)-4, Q&A-3(c), the amount transferred is treated as a benefit paid from the transferor plan.

- (c) Formerly Affiliated Plans of the Employer: A formerly affiliated plan of an employer shall be treated as a plan maintained by the employer, but the formerly affiliated plan shall be treated as if it had terminated immediately prior to the cessation of affiliation with sufficient assets to pay Participants' benefit liabilities under the Plan and had purchased annuities to provide benefits.
- (d) Plans of a Predecessor Employer: If the employer maintains a defined benefit plan that provides benefits accrued by a Participant while performing services for a predecessor employer, the Participant's benefits under a plan maintained by the predecessor employer shall be treated as provided under a plan maintained by the employer. However, for this purpose, the Plan of the predecessor employer shall be treated as if it had terminated immediately prior to the event giving rise to the predecessor employer relationship with sufficient assets to pay Participants' benefit liabilities under the Plan, and had purchased annuities to provide benefits; the employer and the predecessor employer shall be treated as if they were a single employer immediately prior to such event and as unrelated employers immediately after the event; and if the event giving rise to the predecessor relationship is a benefit transfer, the transferred benefits shall be excluded in determining the benefits provide under the Plan of the predecessor employer.
- (e) Special Rules: The limitations of this Article shall be determined and applied taking into account the rules in Treas. Reg. Section 1.415(f)-1(d), (e) and (h).

(f) Aggregation with Multiemployer Plans:

- (i) If the employer maintains a multiemployer plan, as defined in Code Section 414(f), and the multiemployer plan so provides, only the benefits under the multiemployer plan that are provided by the employer shall be treated as benefits provided under a plan maintained by the employer for purposes of this Article.
- 10.8 Participant Limitation Applicable to Deferred Contribution Option: If the Defined Contribution Option is elected in Section 12 of the Joinder Agreement the maximum permissible amount which may be contributed or allocated to or made with respect to any Participant which amount shall be the lesser of:
- (1) \$40,000, as adjusted for cost-of-living under Code Section 415(d) (the "Defined Contribution Dollar Limitation"), or
- (2) 100% of the Participant's Actual Compensation for the Limitation Year.

Notwithstanding any provision of the Plan to the contrary, if the annual additions (within the meaning of Code §415) are exceeded for any participant, then the Plan may only correct such excess in accordance with the Employee Plans Compliance Resolution System (EPCRS) as set forth in Revenue Procedure 2006-27 or any superseding guidance, including, but not limited to, the preamble of the final §415 regulations.

10.9 Re-employment of Former Employees: If a Participant's employment is terminated before he is eligible for a Pension and the Participant is subsequently re-employed by the Employer, the Participant shall only receive credit for his previous period of employment if he did not receive a distribution of Contribution Accumulation, except as otherwise provided under Article VIII,-Employment Transfers. Such an Participant not so entitled to credit for such previous period of employment shall be treated in the same manner as a person who was not previously in the employment of any Municipality.

If a Participant's employment is terminated before he is eligible for a Pension, the Plan provided for the Non-Contribution Option on or before such termination, and the Participant is subsequently re-employed by the Employer, then the Participant will be entitled to credit for such prior Service for benefit accrual purposes, but not for vesting purposes.

10.10 Re-employment of Retired Participants:

- (a) Deferred Vested Retired Participant: If a former Participant retired under this Plan is re-employed by the Employer, and again becomes an Employee under the Plan, no Pension payments shall be made during the period of such re-employment. Upon the subsequent termination of such Employee's employment, the Employee shall be entitled to receive a Pension the amount of which is computed on the basis of the Plan's benefit formula and the Employee's Compensation and Service with the Employer during the period of his re-employment; provided, however, if the Plan's benefit formula was reduced during his absence, the Employee shall instead be entitled to receive a Pension the amount of which is computed on the basis of his Compensation and Service with the Employer prior to the date of his previous retirement, as well as his Compensation and Service with the Employer during the period of his reemployment.
- (b) Retired Participants in Pay Status: In the case of re-employment of a retired Participant who received any Pension payments prior to his re-employment, the Pension payable upon his subsequent Retirement shall be equal to the sum of (i) and (ii), as follows:
 - (i) the same amount he had been receiving for his prior Retirement, adjusted for any applicable cost-of-living adjustments, payable under the same form of annuity elected for his prior Retirement. If the amount was originally paid early, its new early reduction shall be recalculated to exclude the months of re-employment with no payments. Only re-employment months prior to the Participant's Normal Retirement Date would be excluded.
 - (ii) an amount determined solely for his Compensation and Service with the Employer after re-employment, payable in the form elected under Article V, VI or VII. The Participant does not have to select the same form of payment as for Subsection (i).
- 10.11 Buyback of Service: If the Employer elects in the Joinder Agreement, then notwithstanding anything to the contrary herein, a Participant who terminates employment with the Employer and receives distribution of his Contribution Accumulation may be recredited with

his service if he repays his Contribution Accumulation with the interest that would have accrued on such amount under the terms of the Plan.

10.12 Loss of Benefits: [Reserved]

10.13 Loss of Benefits for Cause: [Reserved]

ARTICLE XI.

Guarantees and Liabilities

- 11.1 Non-Guarantee of Employment: Nothing contained in this Plan shall be construed as a contract of employment between the Employer and any Employee, or as a right of any Employee to be contained in the employment of the Employer, or as a limitation of the right of the Employer to discharge any of its Employees, with or without cause.
- 11.2 Rights to Fund Assets: No Participant shall have any right to, or interest in, any assets of the Fund upon termination of his employment or otherwise, except as provided from time to time under this Plan, and then only to the extent of the benefits payable to such Participant out of the assets of the Fund. All payments of benefits as provided for in this Plan shall be made solely out of the assets of the Fund and neither the Employer, the Trust Service Provider, the Authorized Agent, nor any individual Trustee shall be liable in any manner.
- 11.3 Non-Alienation of Benefits: The Fund shall be exempt from legal process and no order may be made to hold, seize, garnishee, or attach payments to any person. Except as duly required under applicable law (including any domestic relations order) benefits payable under this Plan shall not be subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, encumbrance, charge, garnishment, execution, or levy of any kind, either voluntary or involuntary, including any liability for alimony or other payments for property settlement or support of a spouse or former spouse, or for any other relative of the Participant, but excluding devolution by death or mental incompetency, prior to being received by the person entitled to the benefit under the terms of the Plan. Except as may be duly required under applicable law, the Fund shall not in any manner be liable for, or subject to, the debts, contracts, liabilities, engagements, or torts of any person entitled to benefits hereunder and none of the unpaid Plan benefits or Trust assets shall be considered an asset of the Participant in the event of his divorce, insolvency, or bankruptcy.
- 11.4 **Disclaimer of Liability:** Neither the Employer, the Trust Service Provider, the Authorized Agent, the Trustees, nor any individual Trustee guarantees the Fund in any manner against loss or depreciation, and they shall not be liable for any act, or failure to act, which is made in good faith pursuant to the provisions of the Plan. The Employer shall not be responsible for any act, or failure to act, of the Trustees or the Trust Service Provider. The Trustees shall not be responsible for any act, or failure to act, of the Employer or the Authorized Agent.
- 11.5 Indemnification of Trustees: The Trustees shall be indemnified from the assets of the Fund against any and all liabilities arising by reason of any act, or failure to act, made in good faith pursuant to the provisions of the Plan, including expenses reasonably incurred in the defense of any claim relating thereto.

11.6 Payments Under a Qualified Domestic Relations Order:

(a) General: The Municipality shall follow the terms of any "Qualified Domestic Relations Order" as defined in Subsection (b) below issued with respect to a Participant where such Qualified Domestic Relations Order grants to an "Alternate Payee" rights in the benefit of the Participant.

- (b) The term "Qualified Domestic Relations Order" means an order issued by the District Court of the State of Oklahoma pursuant to the domestic relations laws of the State of Oklahoma which relates to the provision of marital property rights to a spouse or former spouse of a Participant and which creates or recognizes the existence of an Alternate Payee's right to, or assigns to an Alternate Payee the right to receive a portion of the benefits payable with respect to a Participant of the Plan.
- (c) To qualify as an Alternate Payee, a spouse or former spouse must have been married to the Participant for a period of not less than thirty (30) continuous months immediately preceding the commencement of the proceedings from which the Qualified Domestic Relations Order issues.
- (d) A Qualified Domestic Relations Order is valid and binding on the Trustees and the Participant only if it meets the requirements of this Section.
 - (e) A Qualified Domestic Relations Order shall clearly specify:
- (1) the name, social security number, and last-known mailing address (if any) of the Participant, and the name and mailing address of the alternative payee covered by the order;
- (2) the amount or percentage of the Participant's benefits to be paid by the Plan to the Alternate Payee;
- (3) the characterization of the benefit as to marital property rights, and whether the benefit ceases upon the death or remarriage of the Alternate Payee; and,
 - (4) each plan to which such order applies.
- (f) A Qualified Domestic Relations Order meets the requirements of this Section only if such order:
- (1) does not require the Plan to provide any type or form of benefit, or any option not otherwise provided under the Plan;
 - (2) does not require the Plan to provide increased benefits; and,
- (3) does not require the payment of benefits to an Alternate Payee which are required to be paid to another Alternate Payee pursuant to another order previously determined to be a Qualified Domestic Relations Order, or an order recognized by the Plan as a valid order prior to the effective date of the Plan.
- (g) A Qualified Domestic Relations Order shall not require payment of benefits to an Alternate Payee prior to the actual retirement date or withdrawal of the related Participant.
- (h) The obligation of the Plan to pay an Alternate Payee pursuant to a Qualified Domestic Relations Order shall cease upon the death of the Participant.
- (i) In the event a Qualified Domestic Relations Order requires the benefits payable to an Alternate Payee to terminate upon the remarriage of said Alternate Payee, the Plan shall

terminate said benefit only upon the receipt of a certified copy of a marriage license, or a copy of a certified order issued by the Court that originally issued said Qualified Domestic Relations Order declaring the remarriage of said Alternate Payee.

- (j) This Section of the Plan shall not be subject to the provisions of the Employee Retirement Income Security Act of 1974 (ERISA), 29 U.S.C.A. Section 1001, et seq., as amended from time to time, or rules and regulations promulgated thereunder, and court cases interpreting said Act.
- (k) The Board of Trustees of the Oklahoma Municipal Retirement Fund shall promulgate such rules as are necessary to implement the provisions of this Section.
- (I) Effective on or after April 6, 2007, a domestic relations order that otherwise satisfies the requirements for a QDRO will not fail to be a QDRO: (i) solely because the order is issued after, or revises, another domestic relations order or QDRO; or (ii) solely because of the time at which the order is issued, including issuance after the annuity starting date.
- (m) An Alternate Payee who has acquired beneficiary rights pursuant to a valid Qualified Domestic Relations Order must fully comply with all provisions of the rules promulgated by the Trustees pursuant to this Section in order to continue receiving his or her benefits.
- (n) Nothing in this Section shall grant a spouse or former spouse of a Participant any property rights in the benefits of any Participant except as specifically authorized for Qualified Domestic Relations Orders, and no spousal consent shall be required for a Participant to elect or change elections pertaining to a benefit payable under this Plan.

ARTICLE XII.

Amendments

- 12.1 Right to Amend: The Employer shall have the right to make from time to time any amendment or amendments to this Plan, in whole or in part, which do not permit reversion of any part of the Fund to the Employer except as provided in Section 13.2 and which do not cause any part of the Fund to be used for, or diverted to, any purpose other than the exclusive benefit of Participants included in the Plan. Any amendments to this Plan, in whole or in part, may be made from time to time by the Employer by ordinance in the same manner as by this original adoption ordinance, but no such amendatory action shall exceed the power and authority granted to the Employer under the laws of the State of Oklahoma.
- 12.2 Amendments: Each Employer agrees to adopt any amendments to this Plan which are necessary for an initial or continued determination that the Plan is a qualified, tax exempt plan under Sections 401(a) and 501(a) of the Code. Any such amendments will be an amendment of the Employer's separate Plan if approved by the Trustee. The Employer may amend its separate Plan in any respect and at any time, subject to the limitations of the Plan, by amendment of or addition to the Joinder Agreement. However, the Oklahoma Municipal Retirement Fund reserves the right to approve all Employer amendments.
- 12.3 Authority of Volume Submitter Practitioner to Amend for Adopting Employers: The Volume Submitter Practitioner (the "Practitioner") will amend the Plan on behalf of all adopting employers, including those employers who have adopted the Plan prior to this amended and restated Plan, for changes in the Code, regulations, revenue rulings, other statements published by the Internal Revenue Service, including model, sample or other required good faith amendments, but only if their adoption will not cause the Plan to be individually designed, and for corrections of prior approved plans. These amendments will be applied to all employers who have adopted the Plan.

The Practitioner will no longer have the authority to amend the Plan on behalf of any adopting employer as of either: (1) the date the Internal Revenue Service requires the employer to file Form 5300 as an individually designed plan as a result of an employer amendment to the Plan to incorporate a type of plan not allowable in the Volume Submitter program, as described in Rev. Proc. 2011-49, or (2) as of the date the Plan is otherwise considered an individually designed plan due to the nature and extent of the amendments. If the Employer is required to obtain a determination letter for any reason in order to maintain reliance on the advisory letter, the Practitioner's authority to amend the Plan on behalf of the adopting employer is conditioned on the Plan receiving a favorable determination letter.

The Practitioner will maintain, or have maintained on its behalf, a record of the employers that have adopted the Plan, and the Practitioner will make reasonable and diligent efforts to ensure that adopting employers have actually received and are aware of all Plan amendments and that such employers adopt new documents when necessary. This Section supersedes other provisions of the Plan to the extent those other provisions are inconsistent with this Section.

ARTICLE XIII.

Termination

- 13.1 Right to Terminate: The Employer may at any time terminate the Plan by proper ordinance and may direct and require the Trustees to liquidate the Fund. If the Employer shall for any reason cease to exist, the Plan shall terminate and the Fund shall be liquidated, unless continued by a successor.
- 13.2 Liquidation of Fund: Upon termination of the Plan or a permanent discontinuance of Employer contributions, the assets of the Fund which have been allocated for the Participants, and interests of the Participants therein as determined by the actuaries, shall be liquidated, after provision is made for the expenses of liquidation, by the payment (or provision for the payment) of benefits accrued prior to the date of termination in the following order of precedence:
- (a) The Contribution Accumulation of each Participant or former Employee entitled thereto under Article VIII, as of the date of such Plan termination, or earlier date of death or Retirement, less other benefit payments, if any, previously received in each case by or on behalf of each such Participant, former Employee, or other eligible Beneficiary. Any such withdrawals on the part of such persons will reduce their interests in distributions under categories (b), (c), (d) and (e) below, on a proportionate basis, as determined by the actuary. Any such person may elect not to take such withdrawals, and have the value thereof included in the actuary's determination of his distributions under categories (b), (c), (d) and (e) below.
- (b) Pensions or other benefits in course of payment to retired Participants, and Beneficiaries of deceased Participants and immediate Pensions for Employees or former Employees entitled thereto under Article VIII, who have reached their Normal Retirement Dates but have not retired.
- (c) Pensions deferred to Normal Retirement Date for Participation who have qualified for an Early Pension.
- (d) Pensions deferred to Normal Retirement Date for Participants who have qualified for a Deferred Vested Pension.

If the funds available in any of categories (b), (c), or (d) are determined to be insufficient to provide all such benefits the funds and benefits shall be apportioned among the various persons, first in category (b), next in category (c), and next in category (d), in the same proportion as each person's accrued credits bears to the accrued credits of all persons in each such category on an Actuarial Equivalent basis as determined by the actuary.

(e) If the cost of providing for the benefits, first in category (b), next in category (c), and next in category (d) is determined to be less than the total funds available, the balance will revert to the Employer.

The benefit any such Participant is entitled to receive under this Plan shall be based on the Participant's Compensation and Service accrued with the Employer prior to the date of termination of the Plan, and his right to such benefit shall be considered as vested to the extent funded, regardless of his age and years of Service on the date of termination of the Plan.

- 13.3 Manner of Distribution: Any distribution after termination of the Plan or permanent discontinuance of Employer contributions, shall be made as soon as administratively feasible, at such times and in such amounts so that no discrimination results, in cash, in securities or other assets in kind (at fair market value), in continued direct payment Pensions, or in nontransferable life insurance or annuity contracts, as the Committee in its discretion, shall determine. In making such distribution, any and all determinations, divisions, appraisals, apportionments and allotments so made shall be final and conclusive and not subject to question by any person.
- 13.4 Consolidation or Merger: Upon the Employer's liquidation, bankruptcy, insolvency, sale, consolidation, or merger to or with another governmental unit in which such Employer is not the surviving unit, the Plan and Fund will terminate and the Fund assets shall be held or distributed as herein provided, unless the successor to the Employer assumes the duties and responsibilities of the Employer by adopting this Plan, or by the establishment of a separate Plan to which the Fund assets shall be transferred with the consent and agreement of the Employer.
- 13.5 Limitations: The order of priorities for distribution set forth above in Section 13.2, in the event of termination of the Plan, shall be subject to (a) the limitations provided in Article X, and (b) such distributions not being determined to be otherwise discriminatory by the Commissioner of Internal Revenue. In the event such either the limitations under Article X become effective or the Commissioner of Internal Revenue rules that the distributions are otherwise discriminatory, adjustments shall be made in the said priorities and amounts of distributions as may be necessary to satisfy the requirements of Article X or of the Commissioner as the case may be.

ARTICLE XIV.

General

- 14.1 USERRA: Notwithstanding any provision of this Plan to the contrary, effective December 12, 1994, contributions, benefits and service credit with respect to qualified military service will be provided in accordance with Section 414(u) of the Code. A Participant returning from military service shall not be entitled to catch-up on Pick-Up Contributions missed during such military service.
- 14.2 Not Contract Between Employer and Participant: Neither the creation of this Plan, nor any amendment to it, nor the creation of any fund, nor the payment of benefits hereunder shall be construed as giving any legal or equitable right to any Participant against the Employer or against the Oklahoma Municipal Retirement Fund, except as provided herein, and all liabilities under this Plan shall be satisfied, if at all, only out of the Fund held by the Oklahoma Municipal Retirement Fund. Participation in the Plan shall not give any Participant any right to be retained in the employ of the Employer, and the Employer hereby expressly retains the right to hire and discharge any Participant at any time with or without cause, as if this Plan had not been adopted, and any such discharged Participant shall have only such rights or interests in the Fund as may be specified herein.
- 14.3 Payment of Fees: The Employer shall pay a fee in an amount determined and revised from time to time by the Oklahoma Municipal Retirement Fund.
- 14.4 Governing Law: The validity, construction and administration of this Plan shall be determined under the laws of the State of Oklahoma.
- 14.5 Counterpart Execution: This Plan may be executed in two or more counterparts, as may be all amendments thereto be executed, and any one of the executed copies shall be deemed an original.
- 14.6 Severability: Every provision of this Agreement is intended to be severable. If any term or provision hereof is illegal or invalid for any reason whatsoever, such illegality or invalidity shall not affect the validity of the remainder of this Plan.
- 14.7 **Number and Gender:** Pronouns and other similar words used herein in the masculine gender shall be read as the feminine gender where appropriate; pronouns and other similar words used herein in the neuter gender shall be read as the masculine or feminine gender where appropriate; and the singular form of words shall be read as the plural where appropriate.
- 14.8 Compensation and Expenses of Administration: If a Trustee, a member of Oklahoma Municipal Retirement Fund, or a member of the Committee is an Employee of the Employer, he shall serve without any additional compensation. The Employer may pay all or part of the expenses of administration of the Plan, including the compensation and expenses of the Trustee, and any other expenses incurred at the direction of the Oklahoma Municipal Retirement Fund, including, without limitation, fees of actuaries, accountants, attorneys, investment managers, investment advisors and other specialists, and any other costs of administering the Plan. To the

extent that any of such expenses are not paid by the Employer, such expenses shall be paid by the Oklahoma Municipal Retirement Fund out of the Fund.

- 14.9 Incorporation of Trust Agreement: The provisions of the Trust Indenture Establishing the Oklahoma Municipal Retirement Fund are incorporated into and made a part of this Plan.
- 14.10 Mistake of Fact: All contributions to the Plan are made subject to the correctness of the amount. In the event a contribution is made to the Plan and Trust by the Employer under a mistake of fact concerning the correctness of such contribution, then the Oklahoma Municipal Retirement Fund shall return such portion of such contribution which is in excess of the amount that would have been contributed had there not occurred a mistake of fact within one year after the payment of the contribution to the Oklahoma Municipal Retirement Fund.

In the case of amounts returned pursuant to this Section 14.10, no earnings attributable to such amounts may be returned to the Employer, but losses attributable thereto shall reduce the amount returned, and no such return shall reduce the balance of any Participant's Municipality Contribution Accounts to less than the balance which would have been credited thereto had such amount not been contributed.

WITNESS WHEREOF, and as conclusive evidence of the adoption of the foregoing instrument comprising the Plan, the Oklahoma Municipal Retirement Fund has caused its corporate seal to be affixed hereto and these presents to be duly executed in its name and behalf by its proper officers thereunto authorized this 27 day of April 2018.

OKLAHOMA MUNICIPAL RETIREMENT FUND



STATE OF OKLAHOMA)	
COUNTY OF OKLAHOMA) }	SS.

Commission # 06008411 Expires 06/01/18

BEFORE ME, the undersigned a Notary Public in and for said County and State, on this 2 day of App. 2018, personally appeared 2014 to me known to be the identical person who subscribed the name of the Oklahoma Municipal Retirement Fund, a municipal corporation, to the foregoing instrument as its Chairperson and acknowledged to me that he executed the same as his free and voluntary act and deed and as the free and voluntary act and deed of such corporation, for the uses and purposes therein set forth.

GIVEN UNDER MY HAND AND SEAL OF OFFICE, the day and year last above written.

Notary Public

My Commission Expires: Ole . 01. Ze

The name, address and telephone number of the Volume Submitter Practitioner are: McAfee & Taft A Professional Corporation, 10th Floor, Two Leadership Square, 211 N. Robinson, Oklahoma City, OK 73102, telephone (405) 552-2231. Any inquiries by the adopting employer regarding the adoption of the Plan, the meaning of Plan provisions, or the effect of the Internal Revenue Service advisory letter on the volume submitter plan may be directed to the Volume Submitter Practitioner.

APPENDIX I .

The following pages contain the actuarial factors needed to determine Actuarially Equivalent options under the Plan.

Oklahoma Municipal Retirement Fund Appendix I LUMP SUM FACTORS BASED ON 5 OR 10 YEAR CERTAIN & LIFE NORMAL FORM

Age	5 Years Cert Immediate	ain & Life Deferred to 55	10 Years Cer Immediate De	tain & Life ferred to 55 De	eferred to 65
00	40 50555	0.04000	10 50040	0.95407	0.33065
20	13.52555	0.84093	13.53249	0.85197	0.35556
21	13.50397	0.90430	13.51132	0.91617	
22	13.48098	0.97246	13.48875	0.98522	0.38236
23	13.45645	1.04576	13.46471	1.05949	0.41119
24	13.43030	1.12462	13.43908	1.13938	0.44219
25	13.40243	1.20944	13.41178	1.22532	0.47554
26	13.37272	1.30068	13.38271	1.31776	0.51142
27	13.34108	1.39884	13.35177	1.41720	0.55002
28	13.30736	1.50444	13.31884	1.52419	0.59154
29	13.27147	1.61805	13.28379	1.63929	0.63621
30	13.23325	1.74029	13.24651	1.76313	0.68427
31	13.19259	1.87182	13.20686	1.89639	0.73599
32	13.14933	2.01337	13.16473	2.03980	0.79164
33	13.10332	2.16569	13.11998	2.19412	0.85154
34	13.05440	2.32965	13.07249	2.36023	0.91600
35	13.00239	2.50612	13.02214	2.53902	0.98539
36	12.94725	2.69614	12.96889	2.73153	1.06011
37	12.88857	2.90069	12.91245	2.93877	1.14053
38	12.82622	3.12092	12.85276	3.16189	1.22713
39	12.76009	3.35809	12.78973	3.40217	1.32038
40	12.69006	3.61356	12.72329	3.66099	1.42083
41	12.61605	3.88883	12.65340	3.93988	1.52907
42	12.53799	4.18554	12.57997	4.24049	1.64573
43	12.45582	4.50549	12.50297	4.56464	1.77153
44	12.36952	4.85066	12.42233	4.91434	1.90725
45	12.27904	5.22323	12.33800	5.29180	2.05374
46	12.18435	5.62560	12.24990	5.69945	2.21196
47	12.08544	6.06045	12.15796	6.14000	2.38293
48	11.98222	6.53067	12.06207	6.61639	2.56782
49	11.87454	7.03944	11.96207	7.13184	2.76787
50	11.76223	7.59025	11.85783	7.68989	2.98444
51	11.64503	8.18689	11.74916	8.29436	3.21904
52	11.52266	8.83351	11.63588	8.94947	3.47329
53	11.39479	9.53467	11.51786	9.65984	3.74898
54	11.26107	10,29538	11.39497	10.43053	4,04809
55	11.12118	11.12118	11.26717	11.26717	4.37279
56	10.97478	10.97478	11.13444	11.13444	4.72548
57	10.82160	10.82160	10,99687	10.99687	5.10882
58	10.66145	10.66145	10.85466	10.85466	5.52586
59	10.49434	10.49434	10.70814	10.70814	5.98006
60	10.32040	10.32040	10.55772	10.55772	6.47547
61	10.13992	10.13992	10.40389	10.40389	7.01677
62	9.95332	9.95332	10.24720	10.24720	7.60941
63	9.76118	9.76118	10.08822	10.08822	8.25984
64	9.56419	9.56419	9.92760	9.92760	8.97568
65	9.36310	9.36310	9.76603	9.76603	9.76603

Oklahoma Municipal Retirement Fund
Appendix I
LUMP SUM FACTORS BASED ON 5 OR 10 YEAR CERTAIN & LIFE NORMAL FORM

	5 Years Certa		10 Years Cert		
Aqe	Immediate I	Deferred to 55	Immediate Def	erred to 55 De	eterred to 65
66	9.15871	9.15871	9.60428	9.60428	9.60428
67	8.95176	8.95176	9.44317	9.44317	9.44317
68	8.74284	8.74284	9.28351	9.28351	9.28351
69	8.53235	8,53235	9.12608	9.12608	9.12608
70	8.32065	8.32065	8.97163	8.97163	8,97163
71	8.10802	8.10802	8.82088	8.82088	8.82088
72	7.89486	7.89486	8.67452	8.67452	8.67452
73	7.68188	7.68188	8.53332	8.53332	8.53332
74	7.47013	7.47013	8.39802	8.39802	8,39802
75	7.26086	7.26086	8.26934	8.26934	8.26934
76	7.05545	7.05545	8.14789	8.14789	8.14789
77	6.85525	6.85525	8.03411	8.03411	8.03411
78	6.66140	6.66140	7.92827	7.92827	7.92827
79	6.47473	6.47473	7.83046	7.83046	7.83046
80	6.29580	6.29580	7.74065	7.74065	7.74065
81	6.12490	6.12490	7.65871	7.65871	7.65871
82	5.96211	5.96211	7.58445	7.58445	7.58445
83	5.80738	5.80738	7.51755	7.51755	7.51755
84	5.66051	5.66051	7.45765	7.45765	7.45765
85	5.52126	5,52126	7.40437	7.40437	7.40437
86	5.38942	5.38942	7.35747	7.35747	7.35747
87	5.26512	5.26512	7.31665	7.31665	7.31665
88	5,14823	5.14823	7.28145	7.28145	7.28145
89	5.03864	5.03864	7.25141	7.25141	7.25141
90	4.93649	4.93649	7.22610	7.22610	7.22610
91	4.84207	4.84207	7.20507	7.20507	7.20507
92	4.75535	4.75535	7.18788	7.18788	7.18788
93	4.67591	4.67591	7.17406	7.17408	7.17406
94	4.60337	4.60337	7.16321	7.16321	7.16321
95	4.53804	4.53804	7.15495	7.15495	7.15495
96	4.47961	4.47961	7.14893	7.14893	7.14893
97	4.42704	4.42704	7.14478	7.14478	7.14478
98	4.38017	4.38017	7.14216	7.14216	7.14216
99	4.33885	4.33885	7.14072	7.14072	7.14072
100	4.30308	4.30308	7.14006	7.14006	7.14006
101	4.27296	4.27296	7.13985	7.13985	7.13985
102	4.24874	4.24874	7.13985	7.13985	7.13985
103	4.23059	4.23059	7.13985	7.13985	7.13985
104	4.21837	4.21837	7.13985	7.13985	7.13985
105	4.21133	4.21133	7.13985	7.13985	7.13985
106	4.20843	4.20843	7.13985	7.13985	7.13985
107	4,20843	4.20843	7.13985	7.13985	7.13985

Basis: .75 (83GAM.M) + .25 (83GAM.F) - 7.5% Interest

Oklahoma Municipal Retirement Fund

Appendix I

LUMP SUM FACTORS BASED ON 5 OR 10 YEAR CERTAIN & LIFE NORMAL FORM

PROCEDURES

- 1. Determine the normal form of annuity for the employer's plan: 5 year certain & life or 10 year certain & life. Only the corresponding columns of Appendix I should be used.
- 2. If the Participant is eligible for immediate payment (under Early, Normal, or Late Retirement) multiply the annual payment amount under the normal form (after any reductions for early payment) by the "Immediate" factor for the Participant's payment start age in completed months. Factors should be interpolated between the two bracketing ages.
- 3. If the Participant is not eligible for immediate payment (Deferred Vested Retirement), multiply the annual payment amount starting at age 65 under the normal form (after any reductions for early payment) by the "Deferred to 55" factor for the Participant's age at the time the lump-sum payment is to be made. Interpolate factors for the Participant's age in completed months.

EXAMPLES

1. Age 60 and 6 months
Normal Form 10 years certain & life
Accrued Benefit \$500.00 / month
Reduced Early Benefit \$387.50 / month

Lump Sum Factor 10.48080

Lump Sum 12 x \$387.50 x 10.48080 = \$48,735.72

2. Age 45 and 6 months

Normal Form 5 years certain & life

Accrued Benefit \$200.00 / month

Reduced Early Benefit at !\$100.00 / month

Lump Sum Factor 5.42442

Lump Sum $12 \times $100.00 \times 5.42442 = $6,509.30$

ORDINANCE NO.	2019-12

AN ORDINANCE AMENDING SECTION 5-5.3(4) OF APPENDIX A TO THE CODE OF ORDINANCES OF THE CITY OF MARIETTA REGARDING THE PLACEMENT OF MOBILE HOMES IN R-1 AND R-2 ZONING LOTS; REPEALING ORDINANCES OR PARTS OF ORDINANCES IN CONFLICT: PROVIDING FOR SEVERABILITY: AND DECLARING AN EMERGENCY

BE IT ORDAINED BY THE MAYOR AND THE CITY COUNCIL OF THE CITY OF MARIETTA, OKLAHOMA:

SECTION 1. Section 5-5.3(4) of Appendix A to the Code of Ordinances of the City of Marietta shall be amended as follows:

- (4) Storage and parking of trailers.
 - a. Individual mobile homes may be permitted to occupy a zoning lot in R-1 or R-2 residential districts upon approval of the planning commission and the city council.
 - b. Prior to permitting a mobile home to occupy a zoning lot in R-1 or R-2 residential districts, the planning commission shall set the matter for public hearing and give at least 15 days' notice of the date, time, and place of the hearing by publication in a newspaper of general circulation and by regular mail to all owners of property within a two hundred (200) foot radius of the subject property, with the addresses of such owners and said radius being as shown by the Office of the Love County Assessor.
 - 1. Protests against proposed changes shall be filed at least three (3) days before the date of the public hearing. If protests are filed by the owners of fifty percent (50%) or more of the area of the lots within a two hundred (200) foot radius of the exterior boundary of the territory included in a proposed change: then the proposed change or amendment shall not become effective except by the favorable vote of three-fourths of all the members of the city council.
 - c. The permit granted by the planning commission, with the approval of the city commission, may be withdrawn at any time if the individual or entity which is permitted to move the individual mobile home upon a lot zoned R-1 or R-2, is not in compliance with the requirements placed on said permit by the planning commission or the city commission council. Any and all requirements shall be prescribed at the discretion of the city commission council but shall include, at a minimum, the following:
 - a-1. The mobile home is compatible with the other residential improvements in the neighborhood as to both value and exterior appearance.
 - b.2. The roof shall be pitched, at a minimum, of 11/2 feet in 12 feet.
 - 1.3. Reserved.

- <u>Written agreement is secured from 75 percent of the owners of property within 200 feet and from all owners of abutting property. Reserved.</u>
- 4.5. The wheels or other transporting devices of all mobile homes permitted in R-1 or R-2 zoned residential districts shall be removed.
- e.<u>6.</u> Any and all mobile homes located in R-1 or R-2 zoned residential districts shall be securely anchored to the ground so as to attain the same resistance to wind as a fixed residence of comparable size.
- £7. As a condition of any permit allowing a mobile home in a R-1 or R-2 zoned residential district, the owner or entity shall be required to skirt the mobile home so as to enclose the underneath area.
- 2.8. Reserved.
- g.-9. The individual or entity which is permitted to move the individual mobile home upon a lot zoned R-1 or R-2 shall have 90 days from the date the mobile home is moved onto the lot to comply with requirements contained in this section. In the event that all requirements are not met within the 90 day period, said permit granted by the planning commission or city commission—council_shall be withdrawn and the mobile home shall be moved within 30 days of notice to the individual or entity placing said mobile home on the lot. Failure to remove said mobile home within the 30-day period shall result in the City of Marietta enforcing this regulation and ordinance by an injunction and court order for said removal, all at the expense of the individual or entity as referred to herein.

SECTION 2. All ordinances or parts of ordinances in conflict herewith are, to the extent of such conflict, repealed.

SECTION 3. If any section, sub-section, sentence, clause, phrase, or portion of this Ordinance is, for any reason, held invalid or unconstitutional by any court of competent jurisdiction, such portion shall be deemed separate, distinct, and independent provision and such holding shall not affect the validity of the remaining portion of this Ordinance.

PASSED by the City Council and SIGNED by the Mayor of the City of Marietta, Oklahoma on this 12th day of November 2019.

ATTEST:

OF MARIETTA

CITY OF MARIETTA

BY:

KIMBERLY FRAIRE, MAYOR

OKLAHOWA

AN ORDINANCE AMENDING THE ZONING OF REAL PROPERTY LOCATED IN THE CITY OF MARIETTA; REPEALNG ORDINANCES OR PARTS OF ORDINANCES IN CONFLICT; AND PROVIDING FOR SEVERABILITY;

WHEREAS, the owner of the real property, located at 1005 Shell Street, hereinafter further described has made application with the City of Marietta requesting the zoning of said real property be amended from C-3 highway commercial and commercial recreation district to R-1 single-family residential district; and

WHEREAS, notice of a public hearing was published in the Marietta Monitor on February 7, 2020 and mailed to the owners of property located within 300 feet of the real property on February 14, 2020; and

WHEREAS, a public hearing regarding the application for zoning amendment of the real property was held before the Planning Commission on February 27, 2020 and the Planning Commission has recommended amendment to the zoning of the real property as provided by Resolution PC 20-03; and

WHEREAS, a public hearing regarding the application for zoning amendment of the real property was held before the City Council on March 10, 2020 and the City Council has determined that the amendment of the zoning district of the real property, located within the corporate limits of the City of Marietta, is in the best interest of the City and the general public.

THEREFORE, BE IT ORDAINED BY THE MAYOR AND THE CITY COUNCIL OF THE CITY OF MARIETTA, OKLAHOMA:

SECTION 1. The zoning of the following property is hereby amended from C-3 highway commercial and commercial recreation district to R-1 single-family residential district to wit:

Lot 18, Block 2, Shellenberger Addition to the City of Marietta, Love County, Oklahoma, according to the recorded plat thereof.

SECTION 2. All ordinances or parts of ordinances in conflict herewith are, to the extent of such conflict, repealed.

SECTION 3. If any section, sub-section, sentence, clause, phrase, or portion of this Ordinance is, for any reason, held invalid or unconstitutional by any court of competent jurisdiction, such portion shall be deemed separate, distinct, and independent provision and such holding shall not affect the validity of the remaining portion of this Ordinance.

PASSED by the City Council and **SIGNED** by the Mayor of the City of Marietta, Oklahoma on this 10th day of March 2020.

CITY OF MARIETTA

ATTEST:

DOTTIE GWIN, CITY CLERK

BY: KIMBERI V FRAIRE MAYOR

ORDINANCE NO. 2020-02

ORDINANCE ENACTING PROVISIONS FOR RIOT PREVENTION AND DISASTER CONTROL IN A STATE OF EMERGENCY AS ALLOWED UNDER THE OKLAHOMA RIOT CONTROL AND PREVENTION ACT; PROVIDING FOR THE FOLLOWING: TITLE; DEFINITIONS; PROCLAMATION AND TERMINATION OF STATE OF EMERGENCY; POWERS OF MAYOR DURING STATE OF EMERGENCY; OFFENSES DURING STATE OF EMERGENCY INCLUDING FOR DESTROYING OR DAMAGING PROPERTY OR INJURING ANOTHER PERSON, FOR ENTRY INTO SCENE OF EMERGENCY, FOR REFUSAL TO LEAVE; FOR LEAVING AN AREA OF QUARANTINE; PROVIDING FOR PENALTY; AMENDMENT OF SECTION 2-174 OF THE MARIETTA CODE OF ORDINANCES; AND REPEAL OF CONFLICTING ORDINANCES; AND DECLARING AN EMERGENCY

WHEREAS, the Oklahoma Riot Control and Prevention Act, which pertains to proclamations by the Governor during a state of emergency, stating what acts may be proclaimed as prohibited, has been enacted by the State of Oklahoma and is codified as Title 21, Oklahoma Statutes, Sections 1321.1 et seq.;

WHEREAS, cities and towns are authorized by Section 1321.9 of said Oklahoma. Riot Control and Prevention Act to enact ordinances in general conformity with the provisions of said Oklahoma Riot Control and Prevention Act:

WHEREAS, it in the best interest of the City of Marietta to enact an ordinance in conformity with the provisions of the Oklahoma Riot Control and Prevention Act, and further to allow the City Council or the Mayor, or both, to declare an emergency with regard to suspending competitive bidding under Section 2-174 of the Marietta Code of Ordinances.

NOW, THEREFORE, BE IT ORDAINED BY THE CITY COUNCIL OF THE CITY OF MARIETTA, OKLAHOMA:

Section 1. Title.

This ordinance shall be known as the State of Emergency Riot Prevention and Disaster Control Ordinance.

Section 2. Definitions.

For the purposes of this ordinance, the definition of "Mayor" shall mean the Mayor of the City of Marietta or the acting Mayor as head of government under Oklahoma law. Further, the term "City" shall mean the City of Marietta.

Section 3. Proclamation and termination of a state of emergency.

- (a) The Mayor, after finding that a public disorder, disaster or riot exists within the City or any part thereof, which affects life, health, property or the public peace, may proclaim a state of emergency in the area affected.
- (b) The proclamation of a state of emergency shall be in writing and shall be signed by the Mayor and shall then be filed with the City Clerk.
- (c) The Mayor shall give as much public notice as practical through the news media of the issuance of a proclamation pursuant to this section.
- (d) The state of emergency shall cease to exist upon the issuance of a proclamation of the Mayor declaring its termination; provided that, the Mayor must terminate the state of emergency when order has been restored in the area affected.

Section 4. Powers of Mayor during a state of emergency.

- (a) The Mayor, with the agreement of the Council President, during the existence of a state of emergency, by proclamation, may, in the area described by the proclamation, which proclamation shall not cover any part or portion of the City not affected by public disorder, disaster or riot at the time the proclamation is issued, prohibit:
 - (1) Any person being on the public streets, or in the public parks or at any other public place during the hours declared by the Mayor to be a period of curfew;
 - (2) Such number of persons, as designated by the Mayor, from assembling or gathering on the public streets, parks or other areas of the City, either public or private;
 - (3) The manufacture, transfer, use, possession or transportation of a Molotov cocktail or any other device, instrument or object designed to explode or produce uncontained combustion;
 - (4) The transporting, possessing or using of gasoline, kerosene or combustible, flammable or explosive liquids or materials in a glass or uncapped container of any kind except in connection with the normal operation of motor vehicles, normal home use or legitimate commercial use;
 - (5) The sale, purchase or dispensing of alcoholic beverages;

- (6) The sale, purchase or dispensing of other commodities or goods, as the Mayor reasonably believes should be prohibited to help preserve and maintain life, health, property or the public peace;
- (7) The use of certain streets, highways or public ways by the public; and
- (8) Such other activities as the Mayor reasonably believes should be prohibited to help preserve and maintain life, health, property, or the public peace.
- (b) In imposing the restrictions provided for by this section, the Mayor, with the agreement of the Council President, may impose them for such times, upon such conditions, with such exceptions and in such areas of this City as the Mayor, from time to time, deems necessary.

Section 5. Offenses during the state of emergency—Generally.

- (a) The provisions of this section shall apply during a state of emergency.
- (b) A person is guilty of riot and an offense when he participates with two or more persons in the course of disorderly conduct:
 - (1) With intent to commit or facilitate the commission of a felony or misdemeanor; or
 - (2) With intent to prevent or coerce official action; or
 - (3) When the accused or any other participant, to the knowledge of the accused, uses or plans to use a firearm or other deadly weapon.
- (c) Any person upon any public way within the described area who is directed by the authorities to leave the public way but refuses to do so shall be guilty of an offense.
- (d) A person is guilty of an offense under this section committed by another person when:
 - (1) Acting with the state of mind that is sufficient for the commission of the offense, he causes an innocent or irresponsible person to engage in conduct constituting the offense; or
 - (2) Intending to promote or facilitate the commission of the

offense he:

- a. Solicits, requests, commands, importunes or otherwise attempts to cause the other person to commit it; or
- b. Aids, counsels or agrees or attempts to aid the other person in planning or committing it; or
- c. Having a legal duty to prevent the commission of the offense, fails to make a proper effort to do so; or
- (3) His conduct is expressly declared by a statute of this state to establish his complicity.
- (e) In any prosecution for an offense under this section in which the criminal liability of the accused is based upon the conduct of another person under this section, it is no defense that:
 - (1) The other person is not guilty of the offense in question because of irresponsibility or other legal incapacity or exemption, or because of unawareness of the criminal nature of the conduct in question or for the accused's criminal purpose, or because of other factors precluding the mental state required for the commission of the offense; or
 - (2) The other person has not been prosecuted for or convicted of any offense based on the conduct in question, or has previously been acquitted thereof, or has been convicted of a different offense or in a different degree, or has legal immunity from prosecution for the conduct in question.
- (f) "Disorderly conduct," as used in this section, means a course of conduct by a person who:
 - (1) Causes public inconveniences, annoyance or alarm, or recklessly creates a risk thereof by:
 - a. Engaging in fighting or in violent, tumultuous or threatening behavior; or
 - b. Making an unreasonable noise or an offensively coarse utterance, gesture or display, or addressing abusive language to any person present; or
 - c. Dispersing any unlawful procession or meeting of persons not being a peace officer of this City and

without lawful authority; or

- d. Creating a hazardous or physically offensive condition which serves no legitimate purpose; or
- (2) Engages with at least one other person in a course of disorderly conduct, as defined in subsection (f)(1) of this section, which is likely to cause substantial harm or serious inconvenience, annoyance or alarm, and refuses or knowingly fails to obey an order to disperse, made by a peace officer to the participants.

Section 6. Destroying or damaging property or injuring another person.

- (a) During a state of emergency, any person who maliciously destroys or damages any real or personal property or maliciously injures another person shall be guilty of an offense.
 - (b) Any person 16 years of age or over, who violates the provisions of this section, shall be prosecuted as an adult.
 - (c) A person is guilty of an offense under this section committed by another person when:
 - (1) Acting with the state of mind that is sufficient for commission of the offense, he causes an innocent or irresponsible person to engage in conduct constituting the offense; or
 - (2) Intending to promote or facilitate the commission of the offense he:
 - a. Solicits, requests, commands, importunes or otherwise attempts to cause the other person to commit it; or
 - b. Aids, counsels or agrees or attempts to aid the other person in planning or committing it; or
 - c. Having a legal duty to prevent the commission of the offense, fails to make a proper effort to do so.
 - (d) In any prosecution for an offense under this section in which the criminal liability of the accused is based upon the conduct of another person pursuant to this section, it is no defense that:
 - (1) The other person is not guilty of the offense in question because of unawareness of the criminal nature of the conduct in question or of the accused's criminal purpose, or because of other

factors precluding the mental state required for the commission of the offense; or

(2) The other person has not been prosecuted for or convicted of any offense based on the conduct in question, or has previously been acquitted thereof, or has been convicted of a different offense or in a different degree, or has legal immunity from prosecution for the conduct in question.

Section 7. Entry into the scene of emergency; refusal to leave; areas of quarantine.

- (a) Any person entering an area declared by Marietta to be the scene of an emergency, which area contains a continuing threat to human life and health, when said person is not a part of the organized rescue efforts controlled by the City of Marietta or the State of Oklahoma or the United States Government, shall be guilty of an offense unless said person is carrying a pass issued by appropriate governmental authority and performing tasks authorized or permitted by said governmental authority.
- (b) After being instructed to leave by authorized emergency personnel, any person refusing to leave an area declared by Marietta to be the site of an emergency which contains an ongoing threat to human life or health shall be guilty of an offense.
- (c) Any person attempting to leave an area which has been declared by Marietta to be an area of quarantine containing an ongoing threat to human life and health through the spread of contagion, infection, or contamination, shall be quilty of an offense.

Section 8. Penalty.

Any person convicted of an offense under this ordinance, or convicted of any violation of the lawful proclamations of the Mayor of the City of Marietta (or authorized designee) during a proclaimed state of emergency, shall be guilty of an offense and upon conviction thereof shall be fined in an amount not exceeding Five Hundred Dollars and No Cents (\$500.00) plus costs. Each such violation and each day of a continuing violation shall be prosecutable as a separate offense.

Section 9. Amendment of Section 2-174.

Section 2-174 of the Code of Ordinances of the City of Marietta is amended as follows:

Sec. 2-174. Emergencies.

limited to conditions resulting from a sudden unexpected happening of unforeseen occurrence or condition and situation wherein the public health; or safety is endangered. The reasons for declaring an emergency and not complying with the sections of this division relating to competitive bidding shall be entered into the official minutes of the City Council.

Section 10. Repeal.

Any ordinances and code provisions in conflict with the provisions of this ordinance are repealed.

Section 11. Codification.

Sections 1 through 8 of this ordinance shall be codified as Article VIII, Sections 70, 400 to 70-407, respectively, under Chapter 70, which is titled "Offenses and Miscellaneous Provisions" of the Marietta City Code.

Section 12. Emergency.

It being immediately necessary for the preservation of the peace, health, safety and public good of the City of Marietta and the inhabitants thereof that the provisions of this ordinance be put into full force and effect, an emergency is declared to exist, and by reason of which this ordinance shall take effect and be in full force from and after the date provided herein as provided by law.

PASSED by the City Council the City Marietta, Oklahoma and SIGNED by the Mayor on this 20th day of March, 2020.

CITY OF MARIETTA:

KIMBERLY FRAIRE, MAYOF

ATTEST:

DOTTIE GWIN, CITY CLER

ORDINANCE	NO.	2021-01
-----------	-----	---------

AN ORDINANCE PROHIBITING OPEN BURNING IN THE CITY OF MARIETTA; PROVIDING FOR EXCEPTIONS; REPEALING ORDINANCES OR PARTS OF ORDINANCES IN CONFLICT; PROVIDING FOR SEVERABILITY; AND DECLARING AN EMERGENCY

BE IT ORDAINED BY THE MAYOR AND THE CITY COUNCIL OF THE CITY OF MARIETTA, OKLAHOMA:

SECTION 1. Chapter 42 of the code of Ordinances of the City of Marietta, is hereby amended to read as follows:

Sec 42-1. Outdoor Burning.

- (a) It is unlawful to burn rubbish, including all kinds of refuse, trash, and garbage, including yard waste, or natural vegetation, or any other combustible material within the corporate limits of the City of Marietta.
- (b) Open fires for cooking and recreation are permitted under the following conditions, and the provisions of sub-section (d) of this section:
 - (1) Fires used for the preparation of food when contained in a metal or masonry appliance designed for such use and when said appliance is used in accordance with manufacturer guidelines;
 - (2) Fires used for recreation when contained in a chiminea, or similar device, and said device is used in accordance with manufacturer guidelines:
 - (3) Bonfires sponsored by an association, partnership, corporation, organization, or group, with a written permit approved by the Fire Chief of the City of Marietta;
 - (4) Fires set for the bona fide purpose of training firefighters with written approval by the Fire Chief of the City of Marietta.
- (c) The burning of trees, brush, and other vegetation is permitted on agricultural zoned properties for land management, land clearing operations, and/or development of land with a written permit approved by the Fire Chief of the City of Marietta, and when burning is conducted in accordance with the provisions of sub-section (d) of this section.
- (d) Any open burn authorized by this Section may be conducted with the following provisions:
 - (1) All fires must be supervised at all times by a person eighteen (18) years of

ORDINANCE NO.	2021-01	Page 1 of 2

age or older;

- (2) The size of the fire does not present a fire danger;
- (3) The fire does not occur during a period where the type of fire is prohibited under an official burn ban:
- (4) No public nuisance is created;
- (5) The fire is controlled so that a traffic hazard is not created as a result of air contaminants being emitted;
- (6) The burning is conducted so that the ambient air quality of the area is not adversely affected.

Sec. 42-2. Penalties.

Any person, firm, or corporation violating any provision of this section shall be guilty of a misdemeanor, and upon conviction, shall be punished by a fine not exceeding five hundred (\$500) dollars, plus costs.

SECTION 2. All ordinances or parts of ordinances in conflict herewith are, to the extent of such conflict, repealed.

SECTION 3. If any section, sub-section, sentence, clause, phrase, or portion of this Ordinance is, for any reason, held invalid or unconstitutional by any court of competent jurisdiction, such portion shall be deemed separate, distinct, and independent provision and such holding shall not affect the validity of the remaining portion of this Ordinance.

PASSED by the City Council and SIGNED by the Mayor of the City of Marietta, Oklahoma on this 8th day of June 2021.

CITY OF MARIETTA

ATTEST:

DOTTIE GWIN, CITY C

SCOTT CHANEY, DEPUTY C

BY

KIMBERLY FRAIRE, MAYOR

ORDINANCE NO. 2021-01

Page 2 of 2